SECURITIES AND EXCHANGE COMMISSION Washington, D.C. 20549

FORM 8-K

CURRENT REPORT

Pursuant to Section 13 or 15(d) of the Securities Exchange Act of 1934

Date of Report: June 15, 1999

RECKSON ASSOCIATES REALTY CORP.
and
RECKSON OPERATING PARTNERSHIP, L.P.
(Exact name of each Registrant as specified in its Charter)

Reckson Associates Realty Corp.-Maryland Reckson Operating Partnership, L.P.-Delaware (State or other jurisdiction of incorporation or organization) Reckson Associates Realty Corp. -11-3233650 Reckson Operating Partnership, L.P. -11-3233647 (IRS Employer ID Number)

1-13762 (Commission File Number)

225 Broadhollow Road
Melville, New York
(Address of principal executive offices)

11747 (Zip Code)

(516) 694-6900 (Registrant's telephone number, including area code)

Item 2. Acquisition or Disposition of Assets

On June 15, 1999, a subsidiary of Reckson Associates Realty Corp. ("Reckson") and Reckson Operating Partnership, L.P. ("Reckson OP"), closed on the acquisition of a first mortgage note secured by 919 Third Avenue in New York City. The mortgage note was acquired from NBBRE-919 Third Avenue Associates, L.P. ("NBBRE") for a purchase price of approximately \$277.5 million. The mortgage note matured in 1997 and is in default. Pursuant to the terms of the mortgage note, a foreclosure sale of the mortgage note may not occur until September 30, 2000 unless a certain type of additional default occurs. The mortgage note entitles its holder to all of the cash flow of 919 Third Avenue and to substantial approval rights with respect to the operation of the property. Reckson will pursue obtaining ownership of 919 Third Avenue, although there can be no assurance as to when or whether Reckson will be able to obtain ownership. If the current owner files a bankruptcy petition, Reckson's ability to exercise its rights over the operation of the property, or to foreclose under the mortgage note, could be delayed or hindered.

The building at 919 Third Avenue is a 42 story, 1.365 million square foot, class A office building. Currently, the building is approximately 98% leased. The law firm of Skadden, Arps, Slate, Meagher & Flom ("Skadden Arps") occupies approximately 705,000 square feet of the building and is scheduled to vacate its space in April 2000. New leases have been signed with Debevoise and Plimpton, a law firm (approximately 417,000 square feet), Schulte, Roth & Zabel, a law firm (approximately 212,000 square feet), and Banque Nationale de Paris, a bank (approximately 188,000 square feet), effectively eliminating the vacancy created by the departure of Skadden Arps.

This transaction was financed through a combination of borrowings under Reckson OP's existing credit facility and the June 2, 1999 private placement sale by Reckson of 6,000,000 shares of Series B Convertible Cumulative Preferred Stock, for aggregate proceeds of \$150 million (the "Series B Preferred Stock"). The terms of Reckson OP's credit facility and the Series B Preferred Stock are described in previous filings made by Reckson and Reckson OP with the Securities and Exchange Commission (the "Commission").

Item 7. Financial Statements and Exhibits

(a) Financial Statements

Financial statements with respect to the acquisition referred to herein, including pro forma financial information, were previously filed with the Commission in a Current Report on Form 8-K filed on or about May 11, 1999.

(c) Exhibits

- 10.1 Consolidated, Amended and Restated Fee and Leasehold Mortgage Note
- 10.2 Endorsement to Note dated June 15, 1999
- 10.3 Agreement of Purchase and Sale, between NBBRE 919 Third Avenue Associates, L.P., as Seller, and Reckson Operating Partnership, L.P., as Purchaser
- 10.4 Side Letter to the Agreement of Purchase and Sale, between NBBRE 919 Third Avenue Associates, L.P., as Seller, and Reckson Operating Partnership, L.P., as Purchaser.
- 99.1 Press Release

SIGNATURES

Pursuant to the requirements of the Securities Exchange Act of 1934, the registrant has duly caused this report to be signed on its behalf by the undersigned thereunto duly authorized.

RECKSON ASSOCIATES REALTY CORP.

By: /s/ Michael Maturo

Michael Maturo
Executive Vice President
and Chief Financial Officer

RECKSON OPERATING PARTNERSHIP, L.P.

By: Reckson Associates Realty Corp., its General Partner

By: /s/ Michael Maturo

Michael Maturo

Executive Vice President and Chief Financial Officer

Date: June 25, 1999

CONSOLIDATED, AMENDED AND RESTATED FEE AND LEASEHOLD MORTGAGE NOTE

\$325,000,000

September, 21, 1993 New York, New York

CONSOLIDATED, AMENDED AND RESTATED FEE AND LEASEHOLD MORTGAGE NOTE (as hereafter may be amended, modified, extended, renewed or substituted, this "Consolidated Note") dated as of September 21, 1993, made by 919 FEE ASSOCIATES L.P. ("Fee Associates") and 919 THIRD AVENUE ASSOCIATES L.P. ("919 Associates"), each a New York limited partnership, and each having an address c/o 919T Corp., The Taylor Simpson Group, One Rockefeller Plaza, Suite 2420, New York, New York 10020 (each a "Borrowing Party", and jointly and severally, the "Borrower") in favor of NBBRE-919 THIRD AVENUE ASSOCIATES, L.P., a Delaware limited partnership, having an address c/o Nomura Babcock & Brown Real Estate, Inc., 1013 Centre Road, Suite 350, Wilmington, Delaware 19805 (the "Lender").

WITNESSETH:

WHEREAS, the Lender has made two mortgage loans to the Borrower in the aggregate principal amount of \$325,000,000 (the "Loans");

WHEREAS, the Loans are evidenced by that certain (a) First Fee and Leasehold Mortgage Note, dated September 30, 1987, from the Borrower to the Lender, in the original principal amount of \$95,000,000, as amended by a Note Modification Agreement, dated as of September 30, 1987, by and among the Lender and the Borrower (the "First Note"), and (b) Modification and Consolidation of Second Fee and Leasehold Mortgage Note and Consolidation, dated as of September 30, 1987, from the Borrower to the Lender, in the original principal amount of \$230,000,000 (the "Second Note");

WHEREAS, (a) the First Note is secured by, among other things, that certain First Fee and Leasehold Mortgage, dated September 30, 1987, from the Borrower to the Lender (the "First Mortgage"), and (b) the Second Note is secured by, among other things, that certain Mortgage Modification and Consolidation of Second Fee and Leasehold Mortgage, Spreader and Consolidation, dated as of September 30, 1987, from the Borrower to the Lender, as amended by a First Amendment to Modification and Consolidation of Second Fee and Leasehold Mortgage, Spreader and Consolidation, dated as of even date herewith, between the Borrower and the Lender (the "Second Mortgage");

WHEREAS, as of the date hereof, the outstanding principal amount of (a) the First Note is \$95,000,000, and (b) the Second Note is \$230,000,000;

WHEREAS, the Lender and the Borrower have entered into that certain Agreement (the "Forbearance Agreement"), dated as of September 1, 1992, regarding, among other things, the interest payment due and owing under the First Note and the Second Note on September 1, 1992, as amended by certain letter agreements;

WHEREAS, the Lender and the Borrower have entered into that certain March Agreement (the "March Agreement"), dated as of March 1, 1993, regarding, among other things, the interest payment due and owing under the First Note and the Second Note on March 1, 1993, as amended;

WHEREAS, the Borrower and the Lender agree that as a result of the current and projected financial condition of the Premises (as hereinafter defined) it is desirable to restructure the Loans;

WHEREAS, as part of the restructuring of the Loans, simultaneously herewith, the Borrower and the Lender have executed and delivered a Consolidated, Amended and Restated Fee and Leasehold Mortgage (as hereinafter may be modified, amended, extended, renewed or substituted, the "Consolidated Mortgage"; capitalized terms used herein and not otherwise defined herein shall have the meaning given such terms in the Consolidated Mortgage) which, among other things, (a) consolidates the lien of the First Mortgage and the Second Mortgage to a single lien in the amount of \$325,000,000, and (b) consolidates, amends and restates the First Mortgage and the Second Mortgage in their entirety; and

WHEREAS, the Lender and the Borrower desire to (a) consolidate the principal amount of the First Note and the Second Note into a single indebtedness in the principal amount of \$325,000,000, and (b) consolidate, amend and restate the First Note and the Second Note in their entirety.

NOW, THEREFORE, Fee Associates and 919 Associates, jointly and severally as the Borrower, covenant and agree as follows:

1. FOR VALUE RECEIVED, 919 Associates and Fee Associates, jointly and severally, HEREBY PROMISE TO PAY on September 30, 1997 (the "Maturity Date"), or the Accelerated Maturity Date (as hereinafter defined), to the order of the

Lender, in same day funds, the principal amount of THREE HUNDRED TWENTY-FIVE MILLION DOLLARS (\$325,000,000) (the "Consolidated Indebtedness"), in lawful money of the United States of America, together with interest thereon, as, and at the rate, provided for herein on any and all principal amounts remaining unpaid hereunder from time to time from and including the date hereof until the date prior to payment in full, plus any other charges or sums payable hereunder (including, without limitation, any Prepayment Fee (as hereinafter defined), the OID Amount (as hereinafter defined), any Yield Maintenance Fee (as hereinafter defined), any Costs and Expenses (as hereinafter defined) and any Additional Interest (as hereinafter defined) or under the Consolidated Mortgage, plus any other amounts due and payable under any of the Loan Documents (collectively, the "Indebtedness"). As provided in the Ancillary Agreement, the Lender is permitted to present the Lender's Sale Notice (as defined in the Ancillary Agreement) requiring the Borrower to either prepay the Indebtedness in full or sell the Premises as provided in Section 2.3 of the Asset Management Agreement.

- 2. The Consolidated Indebtedness evidences (i) the indebtedness of \$95,000,000, as previously evidenced by the First Note and secured by the First Mortgage, and (ii) the outstanding indebtedness of \$230,000,000, as previously evidenced by the Second Note and secured by the Second Mortgage, which amount includes certain other consolidated indebtedness as more particularly described in the Second Note. This Consolidated Note consolidates, amends and restates, and as consolidated, amended and restated, replaces, the First Note and the Second Note in their entirety. This Consolidated Note is not in payment, novation, satisfaction or cancellation of the First Note and the Second Note or of the indebtedness evidenced thereby, such indebtedness being herein ratified and confirmed. It is expressly understood and agreed that this Consolidated Note is given in replacement for the First Note and the Second Note and that no part of the indebtedness evidenced by the First Note and the Second Note shall be disturbed, discharged, cancelled or impaired by the execution and delivery of this Consolidated Note, it being the intention of the Borrower and the Lender that the execution and delivery of this Consolidated Note shall create no new or further principal indebtedness.
- 3. This Consolidated Note is the "Consolidated, Amended and Restated Fee and Leasehold Mortgage Note" secured by, and entitled to the benefits of, inter alia, the Consolidated Mortgage, all of the terms, covenants, conditions and agreements of the Consolidated Mortgage being made a part hereof by this reference.
- 4. In addition to the terms defined above or otherwise herein, the terms set forth below shall have the following meanings:
- (a) "Accelerated Maturity Date" means the earlier to occur of (i) the date on which the holder of this Consolidated Note accelerates the Maturity Date following an Event of Default, (ii) the expiration of the ten Business Day period referred to in the first sentence of Section 2.3 of the Asset Management Agreement, if the Borrower elects to prepay the Indebtedness in full under such Section 2.3 and fails to prepay \$5,000,000 of the Indebtedness within such 10 Business Day period, (iii) the expiration of the 10 Business Day period referred to in Section 2.3 of the Asset Management Agreement, if the Borrower has elected or is deemed to have elected to sell the Premises pursuant to Section 2.3 of the Asset Management Agreement and the Borrower has failed to accept an offer or execute and deliver a sale agreement in compliance with the terms of such Section 2.3, or (iv) the expiration of the ninety day period referred to in the second sentence of Section 2.3 of the Asset Management Agreement, if the Borrower elects to prepay the Indebtedness in full under such Section 2.3 and has failed to prepay the Indebtedness in full as required thereunder.
- (b) "Accrual Period" means the period commencing on March 2, 1992 and continuing through and including the date (the "Note Payment Date") on which the Indebtedness shall have been paid in full, including, without limitation, in connection with a completed foreclosure sale pursuant to which the proceeds thereof shall have been paid to the Lender to satisfy the Indebtedness in full.
- (c) "Additional Interest" means an amount equal to the excess, if any, of (I) forty-nine percent (49%) of the amount, if any, by which the Full Amount (as hereinafter defined) exceeds the Indebtedness (which shall be computed solely for this purpose by excluding Additional Interest and the OID Amount and by computing the aggregate amount payable pursuant to clause (C) of Section 16 and clauses (i), (ii) and (iii) of the third sentence of Section 12(a) hereof if such amounts were computed without regard to the OID Amount) over (II) the excess, if any, of the OID Amount over the aggregate amount that would be payable pursuant to clause (C) of Section 16 and clauses (i), (ii) and (iii) of the third sentence of Section 12(a) hereof as if such amounts were computed without regard to the OID Amount. The Full Amount shall mean either: (i) in the event of a sale, a master lease or other disposition of all or substantially all of the Premises prior to the payment in full of the Indebtedness, the disposition price if the Premises are transferred pursuant to Section 2.3 of the Asset Management Agreement, net of an amount, equal to the reasonable and customary expenses paid by the Borrower in connection therewith; or (ii) if there is no such disposition of the Premises pursuant to clause (i) above, the greater of the disposition price of the Premises (net of

an amount equal to the reasonable and customary expenses paid by the Borrower in connection therewith), if any, and the Appraised Value of the Premises (net of an amount equal to the reasonable and customary expenses the Borrower would pay for a disposition at the price equal to the Appraised Value of the Premises); provided, however, the Lender at any time may elect to waive the determination of the Appraised Value of the Premises, in which event, the relevant amount in clause (ii) above will be such disposition price (net of an amount equal to any reasonable and customary expenses paid by the Borrower in connection therewith). The Appraised Value of the Premises shall mean the appraised value of the Premises as of the date two months prior to the Maturity Date or as of the Accelerated Maturity Date, as the case may be, determined as set forth below:

The Appraised Value of the Premises shall be determined by one MAI appraiser selected by the Lender and one MAI appraiser selected by the Borrower, each of reputable standing in the New York City area. In the event the determinations of the Appraised Value of the Premises made pursuant to the appraisals of the two appraisers so selected differ by an amount not greater than ten percent (10%) of the lower of the two determinations, the Appraised Value of the Premises shall be the average of such two appraisals. In the event such two determinations differ by an amount greater than ten percent (10%) of the lower of such two determinations, the two appraisers so selected shall select a third MAI appraiser of reputable standing in the New York City area (except that if within thirty (30) days of the latter of the date of submission of their determinations they do not select a third appraiser, either the Lender or the Borrower may apply to the American Arbitration Association for appointment of a third MAI appraiser), and the Appraised Value of the Premises shall be the average of all three of the determinations pursuant to such three appraisals except if such third appraiser makes a determination which is higher than the higher of the two initial determinations or lower than the lower of the two initial determinations, in which event the Appraised Value of the Premises shall be the average of the two appraisals closest in dollar amount unless the average of any two appraisals shall be equal to the amount of the remaining appraisal, in which case the Appraised Value shall be the average of the three appraisals (i.e., the middle appraisal). If either party fails to select such an MAI appraiser within thirty (30) days prior to the date of the disposition of the Premises, then, for purposes of determining the Appraised Value, the other party's MAI appraiser's Appraised Value shall govern.

The parties hereto acknowledge and agree that Additional Interest is deemed to be interest and not principal hereunder.

- (d) "Affiliate" means with respect to any person or entity, any other person or entity directly or indirectly controlling or controlled by or under direct or indirect common control with such specified person or entity, any officer, director, shareholder or partner of such specified person or any company of which such specified person or entity is an officer, director, shareholder or partner. For the purposes hereof, the term "control" (including "controlling", "controlled by" and "under common control with") means the possession, direct or indirect, of the power to direct or cause the direction of the management and the policies of a person or entity, whether through the ownership of voting securities, by contract, or otherwise.
- (e) "Approved Operating Budget" means the budget for each calendar year, which budget shall be in the form attached hereto and made a part hereof as Schedule "A", it being understood that the initial Approved Operating Budget which is set forth on Schedule "A" is approved. No later than November 1 of each succeeding calendar year the Borrower shall submit to the Lender a proposed budget. If the Lender, in its reasonable discretion, shall not approve all or any portion of any budget, the Borrower promptly shall revise such budget in accordance with the Lender's requests and shall resubmit such budget or portions thereof to the Lender for its approval. In no event shall any budget be effective prior to the receipt by the Borrower of the Lender's written approval, which approval shall not be unreasonably withheld or delayed. If final approval of all or any portion of any proposed budget has not been given prior to the commencement of the period to which it relates, then, in such event, until a budget or any portion of a budget shall have been approved, the Approved Operating Budget (or applicable

portion thereof) for the immediately preceding calendar year increased or decreased to the extent of any increase or decrease in the CPI Index (as hereinafter defined) in effect on the last day of the calendar year (the "Measurement Year") immediately preceding the calendar year which is the subject of the budget from the CPI Index in effect on the last calendar day of the year immediately preceding the Measurement Year.

The Lender, at any time and from time to time, may give the Borrower notice requiring the Borrower to modify, amend or supplement the Approved Operating Budget then in effect consistent with such notice. The Borrower shall modify, amend or supplement the Approved Operating Budget consistent with the Lender's notice and thereafter submit such new Approved Operating Budget to the Lender for its approval. Once the Lender approves such new Operating Budget, the Approved Operating Budget shall be deemed modified, amended or supplemented, as the case may be.

- (f) "Available Cash Amount" means an amount equal to the excess, if any, of (i) any and all funds being held by or on behalf of the Borrower or either Borrowing Party, including, without limitation, any funds remaining in the Collateral Account as of the Measuring Date (as hereinafter defined) in question, over (ii) the Reserve Amount (as hereinafter defined), any amounts then being held specifically for restoration of the Premises (after any casualty, destruction, damage, condemnation or otherwise), and any then remaining special Expense Amount.
- (g) "Borrower's Partners" means any partner in either Borrowing Party.
- (h) "Business Dav" means any day of the year other than Saturday, Sunday and those holidays on which commercial banks located in the City of New York are required or authorized to close.
- (i) "CPI Index" means the "Consumer Price Index" published monthly by the Bureau of Labor Statistics of the U.S. Department of Labor, All Items, New York, Northern N.J. Long Island, NY-NJ-CT (1982-84=100) for urban wage earners and clerical workers, or a successor or substitute appropriately adjusted. In the event that the CPI Index ceases to use 1982-84=100 as the basis of calculation, or if a substantial change is made in terms or number of items contained in the CPI Index, then the CPI Index shall be adjusted, as determined by the Lender in its reasonable discretion, to the figure that would have resulted had the manner of computing the CPI Index in effect on the date hereof not been altered. In the event such Price Index (or successor or substitute index) is not available, a reliable governmental or other non-partisan publication evaluating the information theretofore used in determining the CPI Index, as determined by the Lender in its reasonable discretion, shall be used.
- (j) "Debt Service Shortfall" means, for any scheduled interest payment due and payable under this Consolidated Note on any Payment Date, the lesser of (i) the amount of such interest payment, and (ii) the excess of (A) the amount of such scheduled interest payment due and payable on such Payment Date under this Consolidated Note, over (B) the Available Cash Amount as of the Measuring Date immediately prior to such Payment Date.
- (k) "Lender's Partners" means any partner in the Lender or any partner in or shareholder of any such partner or any other direct or indirect partner or shareholder which is an owner of a direct or indirect legal or beneficial interest in the Lender or any Affiliate of any of the foregoing, whether such person or entity is, was or shall be one of Lender's Partners (and its or his successors and assigns), or whether such person or entity is known, unknown, disclosed or undisclosed, or is, was or shall be a direct or indirect partner, former partner, employee, officer, director, attorney-in-fact or agent of the Lender or any of Lender's Partners (and each of his or its successors and assigns), or who or which has, had or shall have a direct interest in the Lender of any of the Lender's Partners, in each case only to the extent relating to the transactions contemplated by the Loan Documents.
- (1) "ILP Group" means (i) any partner in either Borrowing Party, or (ii) any partner in or shareholder of any such partner or any other direct or indirect partner or shareholder who or which is an owner of a direct or indirect legal or beneficial interest in Borrowing Party, but only to the extent that such person or entity is himself or itself a signatory to this Agreement or has expressly authorized, in writing, a signatory to this Consolidated Note to bind such non-signatory with respect to the provisions of this Consolidated Note applicable to the LP Group. The term "ILP Group" shall not be deemed to include either Borrowing Party.
- (m) "Measuring Date" means two (2) Business Days prior to each Payment Date (as hereinafter defined).
- (n) "Note Payment Date" shall have the meaning set forth in Paragraph 4(b) hereof.
- (o) "OID Amount" means \$26,764,990.00, it being understood that no interest shall accrue on the OID Amount and the OID Amount, as may be

reduced pursuant to the terms hereof, shall be paid on the Note Payment Date.

- (p) "Outstanding Debt Service Shortfalls" means the aggregate of all accrued and unpaid Debt Service Shortfalls as at any given point in time, it being understood that "Outstanding Debt Service Shortfalls" shall not include interest on Debt Service Shortfalls.
- (q) "Payment Date" means (i) for any period through and including September 30, 1997, the date on which a scheduled payment of interest is due and payable under this Consolidated Note and the Maturity Date if the Note Payment Date is the Maturity Date, and (ii) for any period after September 30, 1997, the last Business Day of February, the last Business Day of August and the Note Payment Date.
- (r) "Prepayment Feel" means a fee equal to five percent (5%) of the principal amount due under this Consolidated Note.
- (s) "Principals" means Harvey Schulweis, Patrick A. Gerschel, Paul E. Taylor, Jr. and Jimbolia Corporation N.V.
- (t) "Reserve Amount" shall mean the amount, from time to time, actually held in the Collateral Account as reserves for the operation of the Premises as set forth in the Approved Operating Budget or as otherwise directed by the Lender in its sole discretion.
- (u) "Yield Maintenance Feel" shall mean a fee, subject to the limitation set forth in Paragraph 17 hereof, equal to the present value of the semi-annual revenue shortfall, if any, determined by subtracting the Liquidation Rate (as hereinafter defined) from the Loan Rate (as hereinafter defined) and multiplying such difference, if positive, by the Principal Liquidation Amount (as hereinafter defined), over the period from the Fee Determination Date through and including the day prior to the Maturity Date (or, thereafter the Note Payment Date), such present value ("PV") to be determined at the Liquidation Rate and calculated, for the semi-annual periods from the Fee Determination Date through and including the day prior to the Maturity Date (or, thereafter, the Note Payment Date), in accordance with the following formula:

$$PV = (P \times (R - T)) \times [(1 - (1 + T)) / T]$$
 where R = the Loan Rate;

- T = the Liquidation Rate;
- n = the number of semi-annual periods from the Fee Determination Date through and including the day prior to the Maturity Date; and
- P = the Principal Liquidation Amount.

For purposes hereof, "Liquidation Rate" means one-half (1/2) of the interest rate per annum equal to the latest three-week moving average of the yields to maturity of U.S. Treasury notes trading closest to par value and maturing on, or within three months of, the Maturity Date, such three-week moving average to be determined by the Lender as of the Fee Determination Date on the basis of yields of such U.S. Treasury notes published by The Wall Street Journal for each of the fifteen (15) Business Days preceding the applicable Fee Determination Date or, if such publication shall be suspended or terminated or such yields shall otherwise not be so published for any such Business Day, on the basis of such yields reported by dealers of U.S. Treasury notes to and published by the Federal Reserve Bank of New York or, if such publication shall be suspended or terminated, on the basis of quotations of such yields received by the Lender from three New York dealers of U.S. Treasury notes of recognized standing selected by the Lender; "Loan Rate" means one-half (1/2) of the Fixed Rate; "Fixed Rate" means "yield to maturity" of the payments of interest scheduled to be made hereunder from and after the "Fee Determination Date" (as defined below) to the Maturity Date; "Fee Determination Date" means the date on which the Loan is repaid; "Principal Liquidation Amount" means that amount of the Loan which is repaid in full.

- 5. Each Borrowing Party acknowledges, agrees and certifies to the Lender that as of the date hereof (a) the Indebtedness is owing under the Loan Documents, without offset, defense, claim or counterclaim of any nature whatsoever, (b) the Indebtedness is secured by the Consolidated Mortgage, which mortgage constitutes, among other things, a validly created and duly perfected first lien on, and security interest in, the Premises as described in the Consolidated Mortgage (subject to the limitations set forth therein), (c) the Loan Documents are unmodified and in full force and effect except as specifically set forth herein, in the Forbearance Agreement and in the March Agreement, and (d) there have been no consents or waivers of any nature whatsoever by the Lender under the Loan Documents, except those consents specifically given in writing.
- 6. Interest shall be due and payable hereunder from the date hereof through and including the Note Payment Date at the following rates per annum during the following loan years:

Loan Year Rate Per Annum

Loan Year 1 through AS PROVIDED IN THE Loan Year 5 (through and including March 1, 1992)

AS PROVIDED IN THE FIRST NOTE AND THE SECOND NOTE

Loan Year 5 (from and after 7.00%

March 2, 1992)

Loan Year 6 7.00%

Loan Year 7 (through and including March 1, 1994)

including March 1, 1994) 7.00%

Loan Year 7 (from and after March 2, 1994)

6.39%

Loan Years 8-10

6.39%

Loan Year 1 will be deemed to have begun on September 30, 1987 and will be deemed to have ended on September 29, 1988, and each subsequent Loan Year shall be deemed to have begun and ended or will begin and end in a comparable manner. Consecutive installments of interest at the above rates or the rates set forth in Paragraph 9 below shall commence on March 1, 1988 and continue on the last Business Day of each August and the last Business Day of each February thereafter occurring before the Note Payment Date and an installment of interest at the above rates or the rates set forth in Paragraph 9 below shall be due on the Maturity Date, the Accelerated Maturity Date or the Note Payment Date, as the case may be. A schedule of interest payments required to be made through and including September 30, 2000 is attached hereto and made a part hereof as Schedule "B".

- 7. Interest on the unpaid principal balance hereof shall be computed and paid for the actual number of days elapsed on the basis of a daily rate based upon the aforesaid interest rate over a period of three hundred sixty (360) days. Interest shall accrue from and including the date hereof through and including the date prior to the Note Payment Date.
- 8. All payments due hereunder shall be made to the Lender or its transferee by wire transfer in immediately available federal funds, or in such other manner as the holder hereof may from time to time designate in writing, and shall be applied by the holder hereof, subject to the terms of Paragraph 15 below, when received, in such order and to such amounts evidenced hereby as the holder hereof at its sole option shall determine, including, without limitation, first to the payment of any charge or sum secured by or payable pursuant to the Consolidated Mortgage (other than principal or interest), next to accrued but unpaid interest, and finally to the reduction of the principal balance hereof; provided, however, that in the event and to the extent that the Lender shall apply such payment other than to interest (including, Outstanding Debt Service Shortfalls), the Lender shall notify the Borrower of such application and the Borrower shall not be deemed to be late with respect to the amount of such application if payment in an amount equal to such application is received by Lender within two (2) Business Days of receipt by the Borrower of such notice. Any payment received by the Lender, as set forth above, after 2:00 PM on any day shall not be deemed for purposes hereof to have been received until the next Business Day.
- 9. The foregoing notwithstanding but subject to the terms of the Consolidated Mortgage, if there shall occur a Basic Default, then, from and after the occurrence of such Basic Default and until the earlier of (a) a cure of such Basic Default has been accepted, in writing and in its sole discretion, by the Lender, or (b) the Note Payment Date, whether or not any action shall have been taken or proceedings commenced to recover any amounts hereunder or to foreclose the Consolidated Mortgage, the Borrower shall pay, subject to the further terms of this Paragraph 9, interest ("Default Interest"), at a fluctuating interest rate per annum (the "Default Rate") equal to the greater of fifteen percent (15%) per annum or four percent (4%) in excess of the rate of interest announced publicly by Citibank, N.A. in New York, New York as its base (or prime) rate on ninety (90) day loans to responsible and substantial commercial customers. Notwithstanding the foregoing to the contrary, with respect to any Basic Default, Default Interest in excess of the stated interest due hereunder shall accrue from the occurrence of such Basic Default and shall be payable at any time on or after the Dispute Determination Date with respect to such Basic Default; provided, however, if (a) the Lender withdraws all of its Basic Default claims pursuant to Section 3.01(B)(vi) of the Consolidated Mortgage, or (b) pursuant to a Determination it is found that either no Basic Default has occurred or that there is a Permitted Default Defense sustained which excuses such Basic Defaults, then, in such event, any accrued Default Interest in excess of stated interest due hereunder with respect to such Basic Defaults shall be deemed not to have accrued and be cancelled. No Default Interest shall be payable with respect to Non-Basic Defaults.
- 10. Nothing in this Consolidated Note or in any other agreement between either Borrowing Party and the Lender shall require the Borrower to pay, or the Lender to accept, interest in an amount which would subject the

Lender to any penalty or forfeiture under applicable law. In no event shall the total of all charges payable hereunder, whether of interest or of such other charges which may or might be characterized as interest, exceed the maximum rate permitted to be charged under applicable law. Should the Lender receive any payment which is or would be in excess of that permitted to be charged under such applicable law, such payment shall have been, and shall be deemed to have been, made in error and shall automatically be held by the Lender as additional cash collateral for the indebtedness evidenced by this Consolidated Note, provided that if the determination of such excess occurs after repayment of this Consolidated Note, such excess shall be returned to the Borrower.

- 11. In the event there is any Event of Default under the Consolidated Mortgage with respect to any payment of interest provided for herein, a late charge of four cents (\$.04) for each dollar of interest not paid in sufficient time to prevent the existence of such Event of Default shall be immediately due and payable to the Lender; provided, however, that with respect to the first semi-annual payment of interest to be overdue, such late charge will be payable only if such payment is overdue for at least two (2) Business Days.
- 12. (a) The Borrower shall not be permitted to prepay the Indebtedness in whole or in part. Each Borrowing Party understands that the Lender and/or the Lender's Partners, have already or may in the future enter into funding or other arrangements for or in connection with the Indebtedness on terms and conditions which could result in substantial losses or adverse or unwanted consequences to the Lender and/or the Lender's Partners if this Consolidated Note or any part thereof is prepaid. In the event of any prepayment (whether voluntary or involuntary, as a result of acceleration due to an Event of Default, or operation of law which constitutes an Event of Default) in respect of the Loans in violation of the foregoing, the Borrower shall be required to pay to the Lender (i) all enforcement costs and expenses under the Consolidated Mortgage ("Costs and Expenses"); (ii) the Yield Maintenance Fee (except as otherwise provided herein or in the Consolidated Mortgage); (iii) the Prepayment Fee (except as otherwise provided herein or in the Consolidated Mortgage); and (iv) Additional Interest (except as otherwise provided herein or in the Consolidated Mortgage); provided, however, that the amounts described in clauses (i), (ii) and (iii) shall be payable only to the extent that the sum of such amounts exceeds the excess of the OID Amount over the amount that would be payable pursuant to clause (C) of Paragraph 16 hereof if such amount were computed without regard to the OID Amount and such amounts shall also be subject to the proviso set forth in Paragraph 16 hereof. Such additional payments pursuant to (ii), (iii) and (iv) above shall compensate the Lender and/or the Lender's Partners for such anticipated losses and consequences without the necessity of the Lender or anyone else establishing the amount of any such losses or consequences, the precise amount thereof being incapable of determination. Any demand(s) by the Lender for Costs and Expenses pursuant to (i) above shall be accompanied by a statement of such Costs and Expenses prepared by the Lender. Payment by either Borrowing Party of the Yield Maintenance Fee, if any, the Prepayment Fee, if any, and Additional Interest, if any, pursuant to (ii), (iii) and (iv) above shall be as liquidated damages and not as a penalty, and shall be made on the Fee Determination Date.
- (b) Any tender of payment of the amount necessary to satisfy the Indebtedness evidenced hereby made after the Accelerated Maturity Date at any time prior to foreclosure or a foreclosure sale, either by either Borrowing Party, their respective successors or assigns or by anyone in behalf of either Borrowing Party, except to the extent otherwise provided in this Consolidated Note or the Consolidated Mortgage, shall be deemed to constitute evasion of the terms hereof and shall be deemed to be voluntary prepayment herein, and such prepayment shall require the payment of the Yield Maintenance Fee, the Prepayment Fee and Additional Interest to the extent provided herein (including this sentence).
- 13. It is expressly agreed that, upon the happening of any Event of Default under the Consolidated Mortgage but subject to the terms of the Consolidated Mortgage, the Lender may accelerate the Loan and notwithstanding the Maturity Date set forth herein the principal sum hereof, together with accrued interest, all interest payable at the Default Rate, if any, the OID Amount, the Yield Maintenance Fee to the extent provided herein, the Prepayment Fee to the extent provided herein, Additional Interest to the extent provided herein, the Outstanding Debt Service Shortfalls, and all other Costs and Expenses, if any, previously incurred and unpaid, shall immediately become due and payable.
- 14. (a) Anything contained in this Consolidated Note, the Consolidated Mortgage or any other Loan Document to the contrary notwithstanding, the Lender by acceptance of this Consolidated Note agrees that, except as provided in Paragraph 14(c) hereof, (i) other than each Borrowing Party, no individual, person or entity, whether such individual person or entity is, was or shall be a partner of either Borrowing Party (and each of his or its successors and assigns), or whether such individual person or entity is known, unknown, disclosed or undisclosed, or is, was or shall be a direct or indirect partner, former partner, employee, officer, attorney-in-fact or agent of either Borrowing Party (and each of his or its

successors and assigns), or who has, had or shall have a direct or indirect interest in either Borrowing Party (and each of his or its successors and assigns) (the persons or entities referred to above being herein sometimes referred to collectively as the "Exculpated Parties"), shall have any personal liability for (A) payment of any of the Indebtedness due under, pursuant to or in connection with any Loan Document, (B) the performance of any of the terms, covenants, or conditions hereof or thereof, or (C) the breach of any covenant, representation or warranty contained in this Consolidated Note, the Consolidated Mortgage, any other Loan Document or in any certificate, notice, financial or other statement, report or abstract heretofore or hereafter delivered, and (ii) in any action to foreclose the Consolidated Mortgage or in any other legal or equitable action arising out of the Consolidated Mortgage, this Consolidated Note or any other Loan Document, or otherwise in any manner whatsoever with respect to any Loan Document, the Exculpated Parties shall not be liable for:

- (aa) any deficiency whatsoever, the Lender hereby covenanting and agreeing that no deficiency or other money judgment will be sought against any Exculpated Party in such action or otherwise in any manner whatsoever,
- (bb) any money judgment or damages arising from any action, suit, proceeding or claim in connection with a breach of any covenant, warranty, representation or undertaking by either Borrowing Party,
- (cc) waste,
- (dd) except as provided in that certain Indemnity Agreement, dated as of even date herewith, by the Principals in favor of the Lender (the "Indemnity Agreement"), the return or refund of any distributions or other payments made to any Exculpated Party prior to the date hereof, or
- (ee) the performance of any of the terms, covenants or conditions of any Loan Document; provided, however, any then general partner of either Borrowing Party shall be liable for such performance so long as either such general partner is reimbursed for its costs and expenses in connection with such performance or such performance does not require such general partner to expend more than any then remaining DeMinimus Amount (as defined in the Consolidated Mortgage),

recourse being strictly limited to the Premises and the other assets of each Borrowing Party (excluding in all events any right or claim of either Borrowing Party against any of the Borrower's Partners or their Affiliates). The Lender for itself, its successors and assigns, by acceptance of this Consolidated Note does hereby agree to look solely to the Premises and such other assets of each Borrowing Party (excluding in all events any right or claim of either Borrowing Party against any of the Borrower's Partners) for the satisfaction of the Indebtedness and all other amounts due under this Consolidated Note, the Consolidated Mortgage or any other Loan Document or for the performance of any of the agreements, obligations, covenants or warranties contained in this Consolidated Note, the Consolidated Mortgage or any of the Loan Documents and no property or assets of any Exculpated Party shall be subject to levy, lien, execution, attachment or other enforcement procedure for the satisfaction of any of the rights and remedies of the Lender.

For purposes of this Paragraph 14 only, the term "Loan Documents" shall be deemed to include that certain Letter Agreement, dated September 30, 1987, to the Lender regarding asbestos and capital improvements at the Premises, which letter agreement is being terminated as of the date hereof pursuant to the Ancillary Agreement.

- (b) Subject to the terms of this Paragraph 14, the Lender by acceptance of this Consolidated Note, further agrees that each of the Exculpated Parties is hereby released from any and all obligations, warranties, guarantees, agreements, claims, actions, causes of actions, costs, losses, liabilities, damages and demands whatsoever, whether foreseen or unforeseen, whether in law or in equity, which the Lender, its successors and assigns ever had, now have or hereafter can, shall or may have for, or in respect of, or by reason of any matter arising out of, this Consolidated Note, the Consolidated Mortgage or any other Loan Documents as heretofore written (i.e., as they were written prior to the execution and delivery of this document).
- (c) (i) Nothing contained above shall be deemed (A) to impair the validity of the Indebtedness as an obligation of the Borrower and each Borrowing Party, or the lien upon or the security interest of the Lender in the Premises, (B) to limit the right of the Lender to name the Borrower or either Borrowing Party or any transferee of any interest in the Premises as a party defendant in any action or suit to foreclose against the Premises or enforce its other remedies under the Loan Documents, (C) to limit the Lender from enforcing its other rights under this Consolidated Note or the Consolidated Mortgage against either Borrowing Party, nor shall the foregoing, (D) constitute a waiver, release or discharge of any of the Indebtedness but

same shall continue until paid or discharged, (E) affect the liability and obligations of certain Exculpated Parties that may arise pursuant to the terms of the Indemnity Agreement, or (F) affect the Lender's rights and remedies under the Security Agreement or the Collateral Account Agreement, and (ii) notwithstanding that they are Exculpated Parties, if any of Borrower's Partners or their Affiliates receives, directly or indirectly, any distribution of funds made by the Approved GP or proceeds thereof, any loan or other payment made by the Approved GP from either Borrowing Party from and after the date hereof except as permitted or contemplated by the Asset Management Agreement or the other Loan Documents, then, in such event, the Approved GP and the person or entity (or any of their Affiliates) who or which received such payment, jointly and severally, shall be personally liable to return such payment to the Borrower, upon written demand by the Asset Manager or the Lender, with interest thereon from the date of the distribution or payment to the date of return at the Prime Rate (as defined in the Indemnity Agreement).

- 15. (a) Notwithstanding the obligations to pay to the Lender interest under this Consolidated Note, the Borrower shall be permitted to accrue a portion of each scheduled payment of interest due and payable under this Consolidated Note on each Payment Date, during the Accrual Period, equal to the Debt Service Shortfall with respect to such Payment Date; provided, however, that, as of such Payment Date, (x) there has been no Determination that a Basic Default has occurred, or (y) a Permitted Default Defense has been sustained which excuses such Basic Default. Without limitation of the foregoing, the accrual rights provided above shall not be affected by any acceleration of the Indebtedness. Notwithstanding any accrual rights hereunder, each Borrowing Party acknowledges and agrees that if after any Payment Date there is a Determination that a Basic Default occurred prior to such Payment Date and there is no Permitted Default Defense sustained which excuses such Basic Default, then, in such event, any Outstanding Debt Service Shortfalls shall be due and payable upon the Dispute Determination Date.
- (b) Interest for all purposes hereunder shall be computed on the basis of a three hundred sixty (360) day year for the actual number of days elapsed.
- (c) If, as of any Payment Date, the Available Cash Amount as of the Measuring Date immediately preceding such Payment Date shall exceed the interest payment due and payable under this Consolidated Note on such Payment Date, then, in such event, the Borrower, if directed by the Lender, in the Lender's sole discretion, shall apply such excess ("Excess Available Cash Amount"), first, to any accrued but unpaid Outstanding Debt Service Shortfalls, and second, either apply the Excess Available Cash Amount to the then Outstanding Debt Service Shortfalls, or retain the Excess Available Cash Amount as part of the Reserve Amount.
- (d) Each Borrowing Party will collect and receive gross receipts and revenue generated by the Premises and pay expenses in accordance with the Approved Operating Budget or as otherwise approved by the Asset Manager or the Lender in due course consistent with ordinary business practices.
- (e) When any term hereof requires interest to be paid at the Default Rate or otherwise until the entire outstanding and unpaid principal balance of this Consolidated Note and all and other sums evidenced hereby or secured by the Consolidated Mortgage shall have been paid in full, then, in such event, unless otherwise provided, such interest shall continue to be due and payable, on demand, at such rates for such period whether or not any action shall have been taken or proceeding commenced to recover the same or to foreclose the Consolidated Mortgage and whether or not judgment has been entered.
- (f) Each Borrowing Party acknowledges and agrees that notwithstanding any acceleration of the Indebtedness as provided under the Loan Documents and notwithstanding, and during the pendency of, any foreclosure or other remedial proceeding brought and continued by the Lender or any other party (including, without limitation, the appointment of a receiver), the Borrower shall pay to the Lender interest under this Consolidated Note as and when interest shall become due and payable as provided herein.
- 16. In addition to all other amounts or charges due and payable by the Borrower to the Lender on the Maturity Date or the Accelerated Maturity Date or the date on which the Borrower may prepay the Loan or such later date following the Maturity Date on which the Indebtedness shall be paid, the Borrower shall pay to the Lender on any such date, as the case may be, (A) the then Outstanding Debt Service Shortfalls, (B) the OID Amount, (C) (i) if the Maturity Date or the Accelerated Maturity Date occurs on or before September 30, 1997, an amount which, when added to the aggregate of the interest payments received and accrued (at the stated rate of interest hereunder, not the Default Rate) during the period commencing after March 1, 1993 through and including the Note Payment Date and the OID Amount will result in the receipt by the Lender of an internal rate of return on \$325,000,000 for such period equal to seven percent (7%) per annum based on a 360 day year for the actual number of days elapsed and calculated by compounding on a semi-annual basis, or (ii) if the Note Payment Date occurs after September 30, 1997, an amount which, when added to the aggregate of the interest payments received and accrued (at the stated rate of interest hereunder, not the Default Rate)

during the period commencing after March 1, 1993 through and including the Note Payment Date and the OID Amount, will result in the receipt by the Lender of the internal rate of return on \$325,000,000, (x) for the period commencing after March 1, 1992 through and including September 30, 1997 equal to seven percent (7%) per annum based on a 360 day year for the actual number of days elapsed or calculated by compounding on a semi-annual basis, and (y) for the period from September 30, 1997 through and including the Note Payment Date, equal to ten and one-half percent (10 1/2%) per annum based on a 360 day year for the actual number of days elapsed and calculated by compounding on a semi-annual basis and (D) all Additional Interest; provided, however, the Lender shall not be entitled to any payment under this Paragraph 16 and Paragraph 12(a)(iv) (other than the OID Amount) in excess of the amount which, when added to all other payments paid or available for payment to the Lender hereunder results in the Lender receiving an internal rate of return on \$325,000,000 for the period from September 30, 1987 to the date used for the calculation of clause (i) or (ii) above, as the case may be, equal to fifteen percent (15%) per annum based on a 360 day year for the actual number of days elapsed and calculated by compounding on a semi-annual basis.

- 17. (a) Each Borrowing Party hereby waives valuation and appraisement, demand, presentment for payment, notice of dishonor, protest and notice of protest of this Consolidated Note.
- (b) Any notice, demand or request relating to any matter set forth herein shall be given in accordance with the terms of the Consolidated Mortgage.
- (c) This Consolidated Note, being executed and delivered in the State of New York, shall be construed and enforced in accordance with (a) the internal laws of the State of New York, and (b) the provisions hereof, without the aid of any canon, custom or rule of law requiring or suggesting construction against the party drafting or causing the drafting in question.
- (d) Time of payment is of the essence as to all dates set forth herein, but subject to the grace periods, if any, set forth herein and in the Consolidated Mortgage, and provided, further, that whenever any payment to be made under this Consolidated Note shall be stated to be due on a day which is not a Business Day, such payment may be made on the next succeeding Business Day and such extension of time shall in such case be included in the computation of payment of interest.
- (e) This Consolidated Note and the terms hereof may not be waived, changed, modified, terminated or discharged orally, but only by an agreement in writing signed by the party against whom enforcement of any waiver, change, modification, termination or discharge is sought.

(f) (INTENTIONALLY BLANK.]

- (g) In no event will there be any payments, charges, premiums, fees and Yield Maintenance, if any, calculated on more than \$325,000,000 of principal indebtedness from Borrower to Lender in connection with this Consolidated Note and the consolidated indebtedness evidenced hereby.
- (h) Each Borrowing Party, on behalf of itself and on behalf of each person or entity claiming by, through or under it (with respect to such claim only) hereby irrevocably and unconditionally (i) submits itself or himself and, to the extent recourse against its or his property is expressly provided by a written agreement executed by such person, its or his property, solely for the purposes of any legal action or proceeding relating to this Consolidated Note or any other Loan Document or for recognition and enforcement of any judgment in respect thereof, to the exclusive jurisdiction of the courts of the State of New York, the courts of the United States of America for the Southern District of New York, and appellate courts thereof (collectively, the "NY Courts"), (ii) consents to the bringing of any such action or proceeding in the New York Courts and waives any objection that it or he may now or hereafter have to the venue of any such action or proceeding in any such court, including, without limitation, any objection that such action or proceeding was brought in an inconvenient court, and agrees not to plead or otherwise assert the same, (iii) agrees to service upon it or him of any and all process in any such action or proceeding at the address set forth in the Consolidated Mortgage, (iv) agrees that nothing herein shall affect the right to effect service of process in any other manner permitted by law, and (v) agrees that a final judgment in any such action or proceeding shall be conclusive and may be enforced in other jurisdictions by suit on the judgment or in any other manner provided by law. The parties hereto agree that any legal action or proceeding relating to this Consolidated Note shall be brought in the NY Courts only; provided, however, that if either Borrowing Party, or any party claiming by, through or under such Borrowing Party breaches or seeks to resist any term, covenant or condition of this Paragraph 17(h), the Lender shall not be bound by the limitations of this sentence with respect to such party breaching or seeking to resist any term, covenant or condition of this Paragraph 17(h).
- (i) Each Borrowing Party on behalf of itself and on behalf of each person or entity claiming by, through or under it, including, without limitation, the Borrower's Partners, the LP Group and the Affiliates of each

of them, hereby irrevocably and unconditionally waive any right to object to the Lender bringing simultaneous actions to foreclose the Consolidated Mortgage, to recover any amounts due under the Indemnity Agreement, or to recover the Indebtedness under this Consolidated Note, the Consolidated Mortgage, the Security Agreement and/or the Collateral Account Agreement, including, without limitation, any rights either Borrowing Party or its direct and indirect partners and their Affiliates, including the Borrower's Partners and LP Group, otherwise may have under Sections 1301 and 1371 of the Real Property Actions and Proceedings Law of the State of New York.

- (j) If, in any action or proceeding, either Borrowing Party or any person or entity claiming by, through or under either of them, including, without limitation, the Borrower's Partners, the LP Group or any Affiliates of any of them, shall assert any claim or counterclaim of any nature whatsoever arising directly or indirectly from the transaction contemplated hereby or any other transaction regarding the Loans, on account of any act or omission occurring before or after the date hereof, then, in any such action or proceeding, each Borrowing Party on behalf of itself and on behalf of any person or entity claiming by, through or under it, including, without limitation, the Borrower's Partners, the LP Group and the Affiliates of each of them, hereby agrees not to assert any such claim or counterclaim against any of the Lender's Partners. Each Borrowing Party on behalf of itself and on behalf of any person or entity claiming by, through or under it, including, without limitation, the Borrower's Partners, the LP Group and the Affiliates of each of them, hereby (i) specifically waives any claim, counterclaim or defense of any nature whatsoever such parties may have against any of the Lender's Partners arising directly or indirectly from the transactions contemplated hereby or any other transaction regarding the Loans, and (ii) agrees not to use or assert any such claim, counterclaim or defense against any of the Lender's Partners in any action or proceeding brought by or against the Lender.
- (k) Each Borrowing Party on behalf of itself and on behalf of any person or entity claiming by, through or under it, including, without limitation, the Borrower's Partners, the LP Group and the Affiliates of each of them, hereby specifically waives and releases any claim, counterclaim or defense of any nature whatsoever such parties may have against the Lender arising directly or indirectly from the transaction contemplated hereby or any other transaction regarding the Loans, but only on account of any act or omission occurring prior to the date hereof.
- (1) Each Borrowing Party acknowledges and agrees that, anything contained in this Consolidated Note or the Loan Documents to the contrary notwithstanding, (i) Lender's Partners shall have no liability whatsoever to the Borrower, its direct and indirect partners and their Affiliates, including the Borrower's Partners and or the LP Group, on account of any act or omission of the Lender occurring before or after the date hereof, and (ii) the Lender's Partners shall have no obligation to make any contribution whatsoever to the Lender on account of any of the Lender's liabilities.
- (m) The Borrower, each Borrowing Party and the Lender acknowledge and agree that (a) during the term of the Loans, the Asset Manager, or its permitted successors and assigns, will be acting on behalf of the Borrower and each Borrowing Party as permitted under the Asset Management Agreement, and (b) the Asset Manager has sole authority to take those actions delegated to it under the Asset Management Agreement and the Lender agrees to accept performance of the terms of any Loan Documents by the Asset Manager for the Borrower's obligations under the Loan Documents; provided, however, nothing contained herein shall be deemed to limit the Borrower's rights against the Asset Manager under the Asset Management Agreement and the Borrower's right to assert a Permitted Default Defense.
- (n) Nothing contained in this Consolidated Note shall constitute, or be construed to constitute, a waiver, either express or implied, by the Lender of any right or remedy that the Lender may have under the Loan Documents, at law or in equity, arising after the date hereof.
- (o) This document may be executed in one or more counterparts, each of which may be executed by one or more of the parties hereto, but all of which, when taken together, shall constitute but one agreement binding on all of the parties hereto.
- (p) For Federal income tax purposes, the parties hereto agree to treat the "issue price" of this Consolidated Note within the meaning of Sections 1273 and 1274 of the Internal Revenue Code of 1986, as amended (the "Code"), as being \$351,764,990.00 of which 26,764,990.00 represents accrued original issue discount with respect to the First Note and the Second Note as of the date thereof) and to treat the interest that is payable hereunder as "adequate stated interest" within the meaning of Section 1274(c)(2) of the Code on the entire issue price (including the portion thereof that represents such accrued original issue discount).
- (q) Any sale, assignment, endorsement or transfer of this Consolidated Note, shall be null and void unless this Consolidated Note is presented or surrendered to the Borrower for registration of the transfer by the Lender or the transferee, duly endorsed in the name of such transferee or

accompanied by written transfer instructions to register such obligation in the name of such transferee, duly executed by the Lender or its duly authorized attorney; provided, however, that no endorsement or other transfer of this Consolidated Note shall be effective if endorsed or transferred to bearer. Upon such presentation or surrender, the Borrower shall be required to reissue this Consolidated Note in accordance with the Lender's endorsement or transfer instructions and to record said transferee as the duly registered holder of this Consolidated Note on the books and records of the Borrower.

IN WITNESS WHEREOF, the parties hereto have executed and delivered this Agreement as of the date first set forth above.

NBBRE - 919 THIRD AVENUE ASSOCIATES, L.P.

By: Nomura Babcock & Brown Real Estate, Inc., its sole General Partner

By: /s/ Kazuhisa Shirakawa

Name: Kazuhisa Shirakawa

Title: Vice President and Assistant Secretary

919 FEE ASSOCIATES L.P.

By: Split Rock Realty Associates, L.P., its General Partner

By: Third Seabury Corp., General Partner

By: /s/ Harvey Schulweis

Harvey Schulweis, President

By: 919G Corp., General Partner

By: /s/ Patrick A. Gerschel

Patrick A. Gerschel,

President

By: 919T Corp., General Partner

By: /s/ Paul E. Taylor, Jr.

Paul E. Taylor, Jr.,

President

By: Jimbolia Corporation N.V.,

Limited Partner

By: /s/ Louis Arovas

Louis Arovas, Vice President

[SIGNATURES CONTINUED ON THE NEXT PAGE]

919 THIRD AVENUE ASSOCIATES L.P.

By: Third Seabury Corp, General Partner

By: /s/ Harvey Schulweis

Harvey Schulweis, President

By: 919G Corp., General Partner

By: /s/ Patrick A. Gerschel

Patrick A. Gerschel,

President

By: 919T Corp., General Partner

By: /s/ Paul E. Taylor, Jr.

Paul E. Taylor, Jr.,

President

By: One Rock Realty Associates, L.P.,

Limited Partner

By: Split Rock Realty Associates, L.P.

By: Third Seabury Corp., General Partner By: /s/ Harvey Schulweis Harvey Schulweis, President By: 919G Corp., General Partner By:/s/ Patrick A. Gerschel Patrick A. Gerschel, President [SIGNATURES CONTINUED ON THE NEXT PAGE] By: 919T Corp., General Partner By: /s/ Paul E. Taylor, Jr. Paul E. Taylor, Jr., President By: Jimbolia Corporation N.V., Limited Partner By: /s/ Louis Arovas Louis Arovas, Vice President

its General Partner

Form of Note Endorsement

ENDORSEMENT TO NOTE
DATED JUNE 15,1999
MADE BY NBBRE-919 THIRD AVENUE ASSOCIATES, L.P.
TO

919 THIRD AVENUE LLC IN THE ORIGINAL PRINCIPAL AMOUNT OF \$325,000,000

PAY TO THE ORDER OF 919 THIRD AVENUE LLC, WITHOUT RECOURSE, COVENANT, REPRESENTATION OR WARRANTY, EXPRESS OR IMPLIED, ORAL OR WRITTEN EXCEPT AS SET FORTH IN, AND SUBJECT TO AND LIMITED BY THE TERMS OF THAT CERTAIN AGREEMENT OF PURCHASE AND SALE, DATED AS OF MAY 10, 1999, AS AMENDED BY LETTER AGREEMENT, DATED AS OF MAY 10, 1999, BY AND BETWEEN THE UNDERSIGNED AND RECKSON OPERATING PARTNERSHIP, L.P., PREDECESSOR-IN-INTEREST TO 919 THIRD AVENUE LLC, AS ASSIGNED TO 919 THIRD AVENUE LLC PURSUANT TO THAT CERTAIN ASSIGNMENT AND ASSUMPTION AGREEMENT, DATED AS OF JUNE 15,1999

NBBRE-919 THIRD AVENUE ASSOCIATES, L.P.

By: NBB Real Estate, Inc. its General Partner

By: /s/ Ichiro Tsunada

Name: Ichiro Tsunada Title: President

Date: June 15, 1999

AGREEMENT OF PURCHASE AND SALE

between

NBBRE-919 THIRD AVENUE ASSOCIATES, L.P.

as Seller,

and

RECKSON OPERATING PARTNERSHIP, L.P.

as Purchaser

Dated as of May ___, 1999

TABLE OF CONTENTS

	Page
ARTICLE I	PURCHASE AND SALE; PURCHASE PRICE1
1.1. 1.2. 1.3.	Purchase and Sale
ARTICLE II	CONDITIONS PRECEDENT TO OBLIGATIONS OF PURCHASER4
2.1. 2.2. 2.3.	Conditions Precedent for Purchaser's Obligations
ARTICLE III	CONDITIONS PRECEDENT TO OBLIGATIONS OF SELLER5
3.1. 3.2.	Conditions Precedent for Seller's Obligations
ARTICLE IV	PURCHASER'S REPRESENTATIONS, WARRANTIES AND COVENANTS6
4.1. 4.2.	Purchaser's Representations and Warranties
ARTICLE V	SELLER'S REPRESENTATIONS, WARRANTIES AND COVENANTS9
5.1. 5.2. 5.3. 5.4.	Seller's Representations and Warranties
ARTICLE VI	CLOSING; EXPENSES AND APPORTIONMENTS18
6.1. 6.2. 6.3. 6.4. 6.5.	Place of Closing
ARTICLE VII	DEFAULTS
7.1. 7.2.	Purchaser Default
ARTICLE VIII	INSPECTIONS; DUE DILIGENCE
8.1. 8.2. 8.3.	Inspection of the Real Property

ARTICLE IX	INDEMNIFICATION
9.1.	Indemnification29
9.2.	Release30
ARTICLE X	CASUALTY AND CONDEMNATION30
10.1.	Casualty30
10.2.	Condemnation31
ARTICLE XI	MISCELLANEOUS
11.1.	No Third-Party Beneficiaries31
11.2.	Broker
11.3.	[Intentionally Blank.]
11.4.	Assignment
11.5.	Waiver of Jury Trial
11.6.	Waiver
11.7.	Captions
11.8.	Prevailing Party33
11.9.	Counterparts
11.10.	Exhibits and Schedules33
11.11.	Notices
11.12.	Construction
11.13.	Amendment
11.14.	Entire Agreement
11.15.	
11.16.	Monetary Amounts
11.17.	
11.18.	Confidentiality34
11.19.	in the state of th
11.20.	
11.21.	
11.22.	Acquisition of Partnership Interests35
LIST OF EXHIB	ITS AND SCHEDULES
Exhibit "A"	- List of Collateral Agreements
Exhibit "A-1"	- List of Ancillary Documents
Exhibit "A-2"	
Exhibit "B"	- Form of Borrower Estoppel
Exhibit "C"	- Form of Assignment and Assumption of Mortgage
Exhibit "D"	- Form of Endorsement of Note
Exhibit "D-1"	
Exhibit "E"	- Form of Assignment and Assumption of Collateral Agreements
Exhibit "F"	- List of Tenants
Exhibit "G"	- Non-Foreign Status Affidavit
Exhibit "H"	- Form of Assignment and Assumption of Share
Exhibit "H-1"	
Exhibit "H-2"	- Share Documents

Approved Budget

Review Materials

Title Letter

ESG Information Book

List of Cash Collateral Accounts

Covenants Exception Schedule

Pre-Closing Expenditures

List of Insurance Policies

List of Borrower Affiliates

Representations and Warranties Exception Schedule

List of Leases; Arrearage Schedule; Leasing Commissions; Security Deposits

List of Service Agreements, Construction Agreements and Construction Contract Information

Schedule "1"

Schedule "2"

Schedule "3"

Schedule "4"

Schedule "5"

Schedule "6"

Schedule "7"

Schedule "8"

Schedule "9"

Schedule "10"

Schedule "11"

Schedule "12"

AGREEMENT OF PURCHASE AND SALE

AGREEMENT OF PURCHASE AND SALE (this "Agreement"), dated as of May _____, 1999, by and between NBBRE-919 THIRD AVENUE ASSOCIATES, L.P., a Delaware limited partnership, having an office c/o NBB Real Estate, Inc., 530 Fifth Avenue, Ninth Floor, New York, New York 10036 ("Seller"), and RECKSON OPERATING PARTNERSHIP, L.P., a Delaware limited partnership, having an office c/o Reckson Associates Realty Corp., 10 East 50th Street, 27th Floor, New York, New York 10022 ("Purchaser").

WITNESSETH:

WHEREAS, Seller is the owner and holder of that certain mortgage loan (the "Loan") (a) evidenced by that certain Consolidated, Amended and Restated Fee and Leasehold Mortgage Note, dated as of September, 21 1993, in the principal amount of \$325,000,000 (the "Note"), made by 919 Fee Associates L.P. and 919 Third Avenue Associates L.P., each a New York limited partnership (collectively, the "Borrower"), jointly and severally, in favor of Seller, and (b) secured by, among other things, that certain Consolidated, Amended and Restated Fee and Leasehold Mortgage, dated as of September, 21 1993, made by the Borrower, as mortgagor, to Seller, as mortgagee, recorded on September 23, 1993 in the Office of the City Register, New York County, in Reel 2010, Page 0041 (the "Mortgage") and encumbering the improved real property (the "Real Property") more particularly described therein; and

WHEREAS, upon and subject to the terms and conditions hereinafter set forth, Seller desires to sell and Purchaser desires to purchase the Loan and the interest of Seller in, to and under the Note, the Mortgage and all instruments, documents and title insurance policies relating to or executed in connection with or pursuant to the Loan as more particularly described on Exhibit "A" annexed hereto and made a part hereof (collectively, the "Collateral Agreements") (the Note, the Mortgage and the Collateral Agreements hereinafter are referred to collectively as the "Loan Documents").

NOW, THEREFORE, in consideration of One Hundred Dollars (\$100) and the mutual covenants and agreements herein contained and other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, Seller and Purchaser hereby agree as follows:

ARTICLE I

PURCHASE AND SALE; PURCHASE PRICE

- 1.1. Purchase and Sale. Subject to and in accordance with the terms and conditions set forth herein, Seller shall sell, assign and transfer the Loan to Purchaser, and Purchaser shall purchase and accept the Loan from Seller, "AS IS", "WHERE IS", "WITH ALL FAULTS", without recourse to Seller and without representation or warranty, express or implied, by Seller, except as specifically set forth herein.
- 1.2. Purchase Price. (a) The purchase price for the Loan is Two Hundred Seventy-Seven Million Five Hundred and Forty-Eight Thousand Dollars (\$277,548,000) (the "Purchase Price") which shall be payable in accordance with the further terms of this Section 1.2.
 - (b) The Purchase Price shall be paid as follows:
- (i) (A) Simultaneously with the execution and delivery of this Agreement, Purchaser shall deliver to Gibson, Dunn & Crutcher LLP ("Escrow Agent"), by wire transfer in immediately available federal funds, the amount of Ten Million Dollars (\$10,000,000) as a deposit (the "Initial Deposit") on account of the Purchase Price.
- (B) Unless Purchaser has previously exercised any Termination Option, on or before the Due Diligence Expiration Date (as hereinafter defined) or such later date as may be set forth in Section 8.3(c) hereof, Purchaser shall deliver to Escrow Agent, by wire transfer in immediately available federal funds, the amount of Five Million Dollars (\$5,000,000) as an additional deposit (the "Additional Deposit") on account of the Purchase Price. Purchaser acknowledges and agrees that upon Purchaser's payment of the Additional Deposit, Purchaser shall be deemed to have forever waived the right to exercise any rights under the terms of Section 8.3 hereof (other than the right to the reduction of the Purchase Price, if applicable). If Purchaser waives or is deemed to have waived it rights under the terms of Section 8.3 hereof and Purchaser fails to pay the Additional Deposit as provided in the first sentence of this clause (B), then, in such event, Purchaser shall be deemed to be in material default of its monetary obligations under this Agreement and Seller shall be entitled to terminate this Agreement and retain the Initial Deposit (and any interest thereon) as provided in Section 7.1 hereof (without regard to any conditions set forth in Section 7.1

- (C) The Initial Deposit and the Additional Deposit, together with all accrued interest thereon shall hereinafter collectively shall be referred to as the "Escrow Funds". The Escrow Funds shall be held in escrow in accordance with the provisions of Section 1.3 hereof. Any interest earned on the Initial Deposit and the Additional Deposit shall be in addition to, and not a credit against, the Purchase Price.
- (ii) At the Closing, Purchaser shall (A) pay to Seller (or as directed in writing by Seller) the balance of the Purchase Price, in the amount of Two Hundred Sixty-Two Million Five Hundred and Forty-Eight Thousand Dollars (\$262,548,000), subject to any apportionments and adjustments for which provision is herein made, and (B) direct the Escrow Agent to deliver to Seller the Escrow Funds.
- 1.3. Escrow Provisions. (a) Escrow Agent shall hold the Escrow Funds in escrow in an interest-bearing account (or as otherwise agreed in writing by Seller, Purchaser and Escrow Agent) in a New York Clearing House Bank or in a nationally recognized "money-market fund" until the Closing or sooner termination of this Agreement and shall pay over or apply the Escrow Funds in accordance with the further provisions of this Section 1.3. If Purchaser shall receive the interest earned on the Deposit, Purchaser shall pay all income taxes owed in connection therewith. The employer identification number of Purchaser is set forth on the signature page hereof. Escrow Agent shall not be liable to Purchaser or Seller for loss occasioned by any deposit of the Escrow Funds made in accordance with this Section 1.3.
- (b) At the Closing, the Escrow Funds shall be paid by Escrow Agent to Seller.
- (c) Subject to, and following in compliance with, the provisions of Section 1.3(e) below, Escrow Agent shall deliver to Seller the Escrow Funds within ten (10) business days following Escrow Agent's receipt of Seller's written demand ("Seller's Demand") for the Escrow Funds stating that Purchaser has defaulted in the performance of Purchaser's obligation to purchase the Loan under this Agreement or that Seller is entitled to use the Escrow Funds to restore the Real Property as provided in Section 8.2(b) hereof (it being understood and agreed that (i) Seller shall have no obligation to restore the Real Property, and (ii) to the extent Seller uses all or any portion of the Escrow Funds to restore the Real Property, Purchaser shall promptly pay to Escrow Agent an amount equal to the Escrow Funds used to restore the Real Property.) Simultaneously with Seller's delivery of Seller's Demand to Escrow Agent, Seller shall deliver a copy of Seller's Demand to Purchaser.
- (d) Subject to, and following in compliance with, the provisions of Section 1.3(e) hereof, Escrow Agent shall deliver to Purchaser the Escrow Funds within ten (10) business days following Escrow Agent's receipt of Purchaser's written demand ("Purchaser's Demand") therefor stating that (i) the Closing did not occur on the Scheduled Closing Date or on such later date to which the Closing shall have been adjourned, as a result of Seller's inability to transfer the Loan in accordance with the provisions of this Agreement, or (ii) Seller is otherwise in material breach under this Agreement. Simultaneously with Purchaser's delivery of Purchaser's Demand to Escrow Agent, Purchaser shall deliver a copy of Purchaser's Demand to Seller.
- (e) If Escrow Agent receives either Seller's Demand or Purchaser's Demand pursuant to and in accordance with Section 1.3 (c) or (d) hereof, as the case may be, then, in such event, prior to releasing the Escrow Funds, Escrow Agent shall deliver a copy of Seller's Demand or Purchaser's Demand, as the case may be, to the non-demanding party within five (5) business days after receipt thereof by Escrow Agent. If Escrow Agent shall have complied with the preceding sentence and shall not have received a written objection to the proposed payment before the tenth (10th) business day following the date of Seller's Demand or Purchaser's Demand, as the case may be, then, in such event, Escrow Agent is hereby authorized and directed to make the payment set forth in such demand. If Escrow Agent shall have received a written objection from either party before such payment, then, in such event, Escrow Agent shall continue to hold the Escrow Funds until otherwise directed by written instructions from both of the parties hereto or by a final judgment of a court of competent jurisdiction; provided, however, that Escrow Agent shall have the right, at any time, to deposit the Escrow Funds with any court of competent jurisdiction and thereby be relieved and discharged of any further obligations under this Agreement. Escrow Agent shall give written notice of any such deposit to Seller and Purchaser. Escrow Agent shall be entitled to rely upon the authenticity of any signature and/or the validity of any writing received by Escrow Agent pursuant to, or otherwise relating to, this Agreement.
- (f) The parties acknowledge and agree that (4) Escrow Agent is acting solely as a stakeholder at their request and for their convenience, (ii) Escrow Agent shall not be deemed to be the agent of either of the parties hereto (provided, however, that the parties hereto acknowledge and agree that (x) Purchaser, by its execution and delivery of this Agreement, has pledged to Seller the Escrow Funds as security for Purchaser's obligations hereunder, (y) Purchaser hereby grants to Seller a first priority lien on, and security interest in, the Escrow Funds, and (z) Escrow Agent shall be deemed to be

Seller's agent for the purposes of such pledge and grant of security interest, (iii) any conflict of interest that may exist because of the Escrow Agent's representation of Seller hereunder is hereby waived, and (iv) Escrow Agent shall not be liable to either of the parties hereto for any act or omission on its part as Escrow Agent. Seller and Purchaser, jointly and severally, shall indemnify, defend and hold harmless Escrow Agent from and against any and all losses, liabilities, costs, claims, damages or expenses (including, without limitation, reasonable attorneys' fees and costs) which may be incurred or suffered by Escrow Agent in connection with the performance of Escrow Agent's duties hereunder other than on account of Escrow Agent's own direct gross negligence or willful misconduct.

ARTICLE II

CONDITIONS PRECEDENT TO OBLIGATIONS OF PURCHASER

- 2.1. Conditions Precedent for Purchaser's Obligations. The following are conditions precedent to Purchaser's obligation to purchase the Loan pursuant to the terms of this Agreement:
- (a) Subject to the terms of Section 5.2 hereof, on the Closing Date (as hereinafter defined), the representations, covenants and warranties of Seller set forth in Section 5.1 hereof shall be true, accurate and complete in all material respects as of the date hereof and as of the Closing Date as though repeated and made anew as of the Closing Date (except that the representation in Section 5.1(j) hereof, shall not be deemed repeated and made anew as of the Closing Date); provided, however, that (i) if any fact and/or information is the subject of or described in Purchaser's Preliminary Notice (as defined in Section 8.3 hereof), then, in such event, Seller shall not be deemed to be in breach (for any purpose under this Agreement) of its representations, covenants and warranties set forth in Section 5.1 hereof if any such fact and/or information would form the basis of such a breach, (ii) if any act or omission of the Borrower (or its direct or indirect partners or their respective affiliates) or condition with respect to the Borrower (or its direct or indirect partners or their respective affiliates), in either case not at the direction or consent of JPMIM or Seller, would result in any breach of any of the representations, covenants or warranties of Seller set forth in Section 5.1 hereof when repeated and made anew as of the Closing Date, then, in such event, Seller shall not be deemed in breach (for any purpose under this Agreement) of its representations, covenants and warranties set forth in Section 5.1 hereof, and/or (iii) if after the date of this Agreement Seller takes any action which is expressly permitted by the terms of this Agreement (other than a general provision to act in accordance with past practices) and such action would result in any breach of the representations, covenants or warranties of Seller set forth in Section 5.1 hereof when repeated and made a new as of the Closing Date, then, in such event, Seller shall not be deemed in breach (for any purpose under this Agreement) of such representations, covenants and warranties set forth in Section 5.1 hereof by reason of such
- (b) Seller shall have materially performed and observed all of its obligations under this Agreement; and $\,$
- (c) At the Closing, Seller shall execute and deliver, or cause to be executed and delivered, the documents and other agreements and instruments required to be delivered by or on behalf of Seller pursuant to the terms of this Agreement.
- 2.2. Termination Option. For the purposes of this Agreement, the term "Termination Option" shall mean the termination of this Agreement by Purchaser pursuant to the terms of Section 6.2(f) hereof, Section 7.2, Section 8.3 or Section 10.1 hereof, which Termination Option shall be exercised by the giving of a written notice of termination from Purchaser to Seller and Escrow Agent. Upon the exercise of the Termination Option by Purchaser, Purchaser shall be entitled to the Escrow Funds, subject to the provisions of Section 1.3 hereof and Section 8.2 hereof, and this Agreement shall be deemed terminated and of no further force or effect and neither party hereto shall have any further rights or obligations hereunder or thereunder, provided, however, that (i) the indemnities contained in Section 4.2, Section 8.2 and Section 11.2 hereof, and (ii) the terms of the Confidentiality Agreement (as hereinafter defined), shall survive any such termination.
- 2.3. Waiver of Conditions. The foregoing conditions are for the benefit only of Purchaser and Purchaser may, in its sole discretion, waive in writing any or all of such conditions and close under this Agreement without any abatement of, or credit against, the Purchase Price. In addition, any act or omission of Purchaser which causes a condition to the performance of Purchaser's obligations hereunder not to be fulfilled shall not constitute the non-fulfillment of such condition.

ARTICLE III

CONDITIONS PRECEDENT TO OBLIGATIONS OF SELLER

3.1. Conditions Precedent for Seller's Obligations. The following are

conditions precedent to Seller's obligation to sell the Loan pursuant to the terms of this Agreement:

- (a) Purchaser shall have paid in full the Purchase Price and otherwise performed, in all but immaterial respects, all of its covenants and agreements contained herein which are required to be performed by it on or prior to the Closing;
- (b) On the Closing Date, the representations, covenants and warranties of Purchaser set forth in Section 4.1 hereof shall be true, accurate and complete in all material respects as of the date hereof and as of the Closing Date, as though repeated and made anew as of the Closing Date; and
- (c) At the Closing, Purchaser shall execute and deliver, or cause to be executed and delivered, the documents and other agreements and instruments required to be delivered by or on behalf of Purchaser pursuant to the terms of this Agreement.
- 3.2. Waiver of Conditions. The foregoing conditions are for the benefit only of Seller and Seller may, in its sole discretion, waive in writing any or all of such conditions and close under this Agreement without any increase in the Purchase Price. In addition, any act or omission of Seller which causes a condition to the performance of Seller's obligations hereunder not to be fulfilled shall not constitute the non-fulfillment of such condition.

ARTICLE IV

PURCHASER'S REPRESENTATIONS, WARRANTIES AND COVENANTS

- 4.1. Purchaser's Representations and Warranties. Purchaser represents, warrants, covenants and agrees with Seller as follows:
- (a) Purchaser is a limited partnership, duly organized, validly existing and in good standing under the laws of the state of Delaware.
- (b) Purchaser has the full power and authority to enter into and consummate all transactions contemplated by this Agreement, has duly authorized the execution and delivery of this Agreement and the consummation of the transactions contemplated hereby, and has duly executed and delivered this Agreement, and all of the obligations of Purchaser hereunder constitute, and, upon execution and delivery by Purchaser of the other documents and instruments to be executed and delivered by Purchaser pursuant hereto, all the obligations of Purchaser thereunder will constitute, legal, valid and binding obligations of Purchaser, enforceable against it in accordance with their respective terms.
- (c) The execution and delivery of this Agreement and the consummation of the transactions contemplated hereby, and the performance of or compliance by Purchaser with the terms and conditions of this Agreement will not (i) conflict with or violate any provisions of Purchaser's organizational documents, (ii) conflict with or violate or result in a breach of any of the provisions of, or constitute a default under, any agreement or instrument to which Purchaser is a party or by which it or any of its property is bound, (iii) conflict with or violate any judgment, order, writ, injunction or decree binding on Purchaser or any of its property, or (iv) conflict with or violate any law, rule, regulation or ordinance applicable to Purchaser or any of its property.
- (d) No approval, authorization, order, license or consent of, or registration or filing with, any governmental authority or regulatory body is required in connection with the execution and delivery by Purchaser of this Agreement and the consummation by Purchaser of the transactions contemplated hereby.
- (e) There is no litigation, claim or proceeding pending or, to Purchaser's knowledge, threatened in writing against Purchaser in any court or before any governmental agency or instrumentality that, if determined adversely to Purchaser, would materially and adversely affect the enforceability of this Agreement or any other document or instrument executed or to be executed in connection herewith or the ability of Purchaser to perform its obligations hereunder or consummate the transactions contemplated hereby.
- (f) Purchaser is solvent and has not filed, nor has there been filed against it, nor do grounds exist for the filing of, any voluntary or involuntary petition in bankruptcy or insolvency and no receiver or trustee or similar custodian has been appointed with respect to its property.
- (g) Purchaser (i) has available to it sufficient funds with which to pay the Purchase Price and to meet its other financial obligations to Seller under this Agreement, and (ii) is not in default under the terms of that certain Credit Agreement, dated as of July 23, 1998, among Purchaser, Reckson Morris Operating Partnership, L.P., the Institutions from time to time party to the Credit Agreement as Lenders and The Chase Manhattan Bank, as Arranger, Book Manager and Administrative Agent, UBS AG, New York Branch as Arranger, Book Manager and Syndication Agent, and PNC Bank, National Association as Documentation Agent. The obligations of Purchaser hereunder are not subject to

any contingency for the benefit of Purchaser regarding the availability of financing (whether secured or unsecured) to provide funds to Purchaser to consummate the transactions contemplated hereby.

- (h) Purchaser is a sophisticated investor and has made its own decision to purchase the Loan in accordance with and subject to the terms and conditions of this Agreement. Purchaser has adequate information concerning the business and financial condition of the Borrower and the Real Property to make an informed decision regarding the purchase of the Loan and has independently and without reliance upon Seller, and based on such information as Purchaser has deemed appropriate, made its own analysis and decision to enter into this Agreement. Purchaser acknowledges that the consideration paid pursuant hereto for the purchase of the Loan may differ both in kind and amount from any payments or distributions which may ultimately be received with respect to the Loan. Purchaser acknowledges that the Loan may have limited or no liquidity and that it has the financial wherewithal to own the Loan for an indefinite period of time and that it is assuming and bearing the credit risk of full or partial loss of all amounts hereunder and all other credit risk which is inherent in the Loan, and all collectibility risks associated therewith. Purchaser represents and warrants that Purchaser has made its own independent evaluation and investigation of the business and financial condition of the Borrower (and its direct and indirect partners and their respective affiliates) and the organizational structure of the Borrower (and its direct and indirect partners and their respective affiliates) and acknowledges and agrees that Purchaser's satisfaction with such matters shall not constitute or be deemed to constitute a condition to Purchaser's obligations hereunder. Purchaser further acknowledges and agrees that if, at any time, the Borrower (or its direct or indirect partners or their respective affiliates) commences or institutes any case, proceeding or other action (i) seeking relief on the Borrower's behalf as debtor, or to adjudicate the Borrower a bankrupt or insolvent, or seeking reorganization, arrangement, adjustment, winding-up, liquidation, dissolution, composition or other relief with respect to the Borrower or its debts under any existing or future law of any jurisdiction, domestic or foreign, relating to bankruptcy, insolvency, reorganization or relief of debtors, or (ii) seeking appointment of a receiver, trustee, custodian or other similar official for the Borrower or for all or any substantial part of its property (or any such case, proceeding or action referred to in clause (i) and/or clause (ii) is commenced or instituted against the Borrower), then, in any such event, such matter or fact shall not constitute or be deemed to constitute a condition to Purchaser's obligations hereunder.
- (i) As of the Closing Date, Purchaser shall have inspected the Loan Documents and the Real Property and, as a result of such inspection, shall be fully familiar with the Loan Documents and the Real Property, the present physical and financial condition of the Real Property, the present state of repair of the Real Property and title to the Real Property. Subject to the terms of this Agreement, at the Closing, Purchaser shall accept the Loan, the Loan Documents and the Real Property "AS IS", "WHERE IS" and "WITH ALL FAULTS" on the Closing Date, without any reduction in the Purchase Price for any change in the physical or financial condition occurring from and after the date $\dot{}$ hereof. Purchaser acknowledges and agrees that (i) neither Seller's Partners (as hereinafter defined), J.P. Morgan Investment Management Inc. ("JPMIM"), Insignia/ESG, Inc. ("ESG"), NBB Real Estate, Inc. ("NBB") and their respective officers, directors, shareholders, direct or indirect partners, employees, agents, brokers, representatives and affiliates of any of the foregoing (collectively, the "Seller Parties") nor any other person (other than Seller as expressly set forth and limited herein) has made any representation, warranty or covenant, express or implied, with respect to Loan, the Borrower, the Loan Documents or the Real Property, the fitness, merchantability, suitability or adequacy of the Real Property for any particular purpose, any environmental condition at or with respect to the Real Property, the site or physical conditions applicable to or with respect to the Real Property, the zoning regulations or other governmental requirements applicable to or with respect to the Real Property, the Leases (as hereinafter defined), title to the Real Property or any other matters affecting the Loan or the use, occupancy, operation, ownership or condition of or with respect to the Real Property, and (ii) neither the Seller Parties nor any other person will have, or be subject to, any liability to Purchaser or any other person resulting from the distribution to Purchaser, or Purchaser's use of, any information pertaining to the Loan which is not specifically set forth in this Agreement. Without limiting the generality of the foregoing, Purchaser further acknowledges and agrees that (x) Purchaser has reviewed and it is fully familiar with the current state of title of the Real Property and no representation, warranty, covenant or indemnity has been made with respect thereto, and (y), except as expressly set forth herein, no representation, warranty, covenant or indemnity has been made or will be given to Purchaser or any other person in respect of any environmental liability with respect to any dangerous, toxic or hazardous wastes, materials, pollutants or substances ("Hazardous Materials"), as such terms are defined in federal, state and/or local environmental laws and regulations, including, without limitation, the United States Comprehensive Environmental Response, Compensation and Liability Act 1980, as amended (collectively, "Environmental Laws"). Purchaser also acknowledges and agrees that in no event whatsoever shall any Seller Parties have any liability to Purchaser, or otherwise, with respect to Hazardous Materials affecting the Real Property or Environmental Laws. Purchaser also acknowledges that as of the Closing Date Purchaser shall have had sufficient opportunity to conduct such

investigations of and with respect to the Loan as it has deemed necessary and advisable. Purchaser's representations set forth in this clause (i) shall survive the Closing.

- (j) Except as specifically set forth in this Agreement, Purchaser has not been induced by, and has not relied upon, any representations, warranty or statement made by any Seller Parties or any person representing any Seller Parties. Purchaser's representations set forth in this clause (j) shall survive the Closing.
- (k) For the purposes of this Agreement, the term "Seller's Partners" shall mean any partner in Seller or any partner in or shareholder of any such partner or any other direct or indirect partner or shareholder which is an owner of any direct or indirect legal or beneficial interest in the Seller or any affiliate of any of the foregoing, whether such person or entity is, was or shall be one of the Seller's Partners (and its or his successors and assigns), or whether such person is known, unknown, disclosed or undisclosed, or is, was or shall be a direct or indirect partner, former partner, employee, officer, director, attorney-in-fact or agent of Seller or any of Seller's Partners (and each of his or its successors and assigns), or who or which has, had or shall have a direct interest in Seller or any of Seller's Partners, in each case only to the extent relating to the transactions contemplated by the Loan Documents.
- 4.2. Covenants of Purchaser. From and after the date hereof until the Closing, Purchaser shall (i) not take any action, grant any waiver or release, or expressly consent to or approve any action, with respect to the Loan, the Borrower or the Loan Documents or any of the other documents held by Seller in connection therewith or in connection with the Real Property which would have the effect of impairing or diminishing the value thereof or the validity, enforceability, perfection or priority of the liens of any Loan Document (it being understood and agreed that Purchaser shall indemnify, defend and hold harmless Seller and the Seller Parties from and against any and all Liabilities (as hereinafter defined) arising out of any breach by Purchaser of the terms of this Section 4.2(i), which indemnity, defense and hold harmless shall survive the Closing or earlier termination of this Agreement), (ii) deliver to Seller copies of all notices given or received by Purchaser in connection with the Real Property and/or the Loan, and (iii) be entitled to contact any contractor, subcontractor, materialman, architect, engineer or consultant providing services or otherwise performing work on the Real Property, provided that each and every such contact shall have been coordinated in advance with Seller's representative, Ray Quartararo.

ARTICLE V

Seller's Representations, Warranties and Covenants

- 5.1. Seller's Representations and Warranties. Except as set forth on Schedule "1" annexed hereto and made a part hereof, Seller, hereby represents, warrants, covenants and agrees with Purchaser as follows:
- (a) Seller is a limited partnership, duly organized, validly existing and in good standing under the laws of the State of Delaware.
- (b) Seller has the full power and authority to enter into and consummate all transactions contemplated by this Agreement, has duly authorized the execution and delivery of this Agreement and the consummation of the transactions contemplated hereby, and has duly executed and delivered this Agreement, and all of the obligations of Seller hereunder constitute, and, upon execution and delivery by Seller of the other documents and instruments to be executed and delivered by Seller pursuant hereto, all the obligations of Seller thereunder will constitute, legal, valid and binding obligations of Seller, enforceable against it in accordance with their respective terms.
- (c) The execution and delivery of this Agreement and the consummation of the transactions contemplated hereby, and the performance of or compliance by Seller with the terms and conditions of this Agreement will not (i) conflict with or violate any provisions of Seller's organizational documents, (ii) conflict with or violate or result in a breach of any of the provisions of, or constitute a default under, any agreement or instrument to which Seller is a party or by which it or any of its property is bound, (iii) conflict with or violate any judgment, order, writ, injunction or decree binding on Seller or any of its property, or (iv) conflict with or violate any law, rule, regulation or ordinance applicable to Seller or any of its property.
- (d) No approval, authorization, order, license or consent of, or registration or filing with, any governmental authority or regulatory body is required in connection with the execution and delivery by Seller of this Agreement and the consummation by Seller of the transactions contemplated hereby.
- (e) There is no litigation, claim or proceeding pending or, to Seller's knowledge threatened in writing against Seller in any court or before any governmental agency or instrumentality that, if determined adversely to Seller, would materially and adversely affect the enforceability of this Agreement or any other document or instrument executed or to be executed in

connection herewith or the ability of Seller to perform its obligations hereunder or consummate the transactions contemplated hereby.

- (f) Seller is solvent and has not filed, nor has there been filed against it, nor do grounds exist for the filing of, any voluntary or involuntary petition in bankruptcy or insolvency and no receiver or trustee or similar custodian has been appointed with respect to its property.
- (g) Seller heretofore has made available for Purchaser's inspection the Loan Documents and certain other documents and instruments executed and delivered from time to time in connection with the Loan as more particularly described on Exhibit "A-1" annexed hereto and made a part hereof (collectively, the "Ancillary Documents").
- (h) Seller has made available for Purchaser's inspection true, correct and complete copies of the Loan Documents and the Ancillary Documents. Except as set forth on Exhibit "A" and "A-1" and except as provided in the Loan Documents and/or the Ancillary Documents, the Loan Documents and the Ancillary Documents have not been amended, modified or otherwise changed in writing by Seller, and there are no other written documents, instruments or agreements material to the understanding of the current status of the Loan which (i) were signed by or on behalf of Seller, (ii) are binding on Seller, and (iii) affects the Borrower's rights and obligations under the Loan. Annexed hereto and made a part hereof as Exhibit "A-2" is a true, correct and complete copy of Seller's approval of the Approved Budget in effect as of the date hereof for 1999. The Mortgage has not been satisfied, or, to Seller's knowledge, subordinated in writing by Seller. Seller has not waived in writing any of its material rights under, or otherwise consented in writing to any material departure from, the terms of the Loan Documents or the Ancillary Documents.
- (i) Seller is the sole legal and beneficial owner of the Loan, free and clear of any lien, claim, security interest, option, equity or other charge or encumbrance of any nature whatsoever. Seller has not endorsed, granted, assigned, transferred or otherwise pledged, encumbered or set over the Note to any other person.
- (j) To Seller's knowledge, Seller has not received from Borrower written notice of any setoff, counterclaim, lender liability claim or defense, or any right of rescission with respect to the Loan. To Seller's knowledge, Seller has not received from Borrower written notice of any adverse claims asserted with respect to the enforceability or priority of the Loan Documents or the Ancillary Documents.
- (k) No claims have been made by Seller under its lender's title insurance policies listed on Exhibit "A-1" hereto (collectively, "Lender's Title Policy"). Seller is the owner of Lender's Title Policy. To Seller's knowledge, Seller has not received written notice of any defenses by any title insurer under Lender's Title Policy.
- (1) The outstanding principal amount under the Note as of the date hereof is \$ 325,000,000 (the "Outstanding Principal Balance") and Outstanding Debt Service Shortfalls (as defined in the Note) as of April 30, 1999 was [redacted] (including the OID Amount). No Additional Interest (as defined in the Note) has been paid. The OID Amount (as defined in the Note) as of the date hereof is [redacted]. Except for the accounts listed on Schedule "2" annexed hereto and made a part hereof (collectively, the "Cash Collateral Accounts") established pursuant to the Collateral Account Agreement listed on Exhibit "A" hereto, there are no escrows or other deposits held by Seller under the Loan Documents. The total balance in the Cash Collateral Accounts as of March 31, 1999 was approximately [redacted].
- (m) Annexed hereto and made a part hereof on Schedule "3" is a true, complete and correct copy of the Approved Budget in effect as of the date hereof for 1999.
- (n) To Seller's knowledge, annexed hereto and made a part hereof on Schedule "4" is a true, complete and correct list of all of the leases, together with all amendments, modifications and supplements thereto and guarantees thereof as of the date hereof (collectively and as amended, modified and supplemented, the "Leases") for space at the Real Property (other than subtenancies, if any, under the Leases) and true, correct and complete copies of the Leases have been made available to Purchaser. To Seller's knowledge, Schedule "4" annexed hereto and made a part hereof is a true, correct and complete list of the security deposits held by the Borrower with respect to the Leases, and the bank(s) and account number(s) where such security deposits are maintained.
- (o) To Seller's knowledge, all of the Leases are in full force and effect (although Purchaser acknowledges and understands that certain Leases have not commenced) and none of the Leases has been further modified, amended or supplemented.
- (p) To Seller's knowledge, (i) there are no pending summary proceedings for the eviction of any tenant, (ii) there are no pending proceedings or pending written claims by any tenant for offsets against rent or additional rent or for damages or other redress, and (iii) no tenant has

delivered written notice to the Borrower, JPMIM or ESG for which such tenant is currently disputing the amount of additional rent or escalation payments due pursuant to such tenant's Lease.

- (q) To Seller's knowledge, neither the Borrower, ESG nor JPMIM has received written notice from any tenant claiming that the Borrower is in default of any of its obligations under any Lease, which default has not been cured.
- (r) To Seller's knowledge, (i) neither JPMIM, ESG nor the Borrower has sent any written notice to any tenant claiming that such tenant is in default, which default remains uncured, and (ii) except as set forth on Schedule "4" hereto, no tenant is (x) in default of the payment of monthly base rent, or (y) otherwise in material breach of a material term of its Lease to such an extent that a prudent landlord of a first class building similar to the Real Property would seek to terminate such Lease as a result of such material breach
- (s) To Seller's knowledge, except as set forth on Schedule "4" hereto, there are no delinquencies in any rental payments due under any of the Leases.
- (t) To Seller's knowledge, Schedule "4" hereto includes a list of all brokerage commissions, finder's fees or real estate agent's fees applicable to the [redacted] (as such Leases are defined in Schedule "4" hereto). To Seller's knowledge, Schedule "4" hereto includes a true, correct and complete list of all brokerage agreements regarding the payment of brokerage commissions, finder's fees or agent's fees applicable to the Leases for which payment has not been made, and true, correct and complete copies of such brokerage agreements have been made available to Purchaser.
- (u) Seller is not a "foreign person" within the meaning of Section 1445 of the Internal Revenue Code, as amended.
- (v) To Seller's knowledge, neither Seller nor the Borrower has received notice of any condemnation or eminent domain proceeding pending or threatened, in writing, against the Real Property or any part thereof.
- (w) To Seller's knowledge, (i) Schedule "5" annexed hereto and made a part hereof is a true, correct and complete list of all material service, utility and maintenance contracts, together with all amendments, modifications and supplements thereto (collectively and as amended, modified and supplemented, the "Service Agreements"), and all construction contracts and architect's and engineer's agreements (together with all amendments, modifications and supplements thereto) relating to work being performed by the Borrower at the Real Property (collectively and as amended, modified and supplemented, the "Construction Agreements"), (ii) the Service Agreements and the Construction Agreements are in full force and effect, without material default by (or notice of material default to) any party, and (iii) true, correct and complete copies of the Service Agreements and the Construction Agreements have been made available to Purchaser. Schedule "5" hereto sets forth, to Seller's knowledge, the percentage of completion of work under the Construction Contracts, amounts paid and balance due thereunder and any retainages with respect thereto, all as of the date hereof.
- (x) To Seller's knowledge, Seller has provided to Purchaser a copy of all environmental reports with respect to the Real Property which are in the possession or control of Seller (collectively, the "Environmental Reports"), excluding, however, those Environmental Reports restricted by confidentiality arrangements (and Seller shall notify Purchaser of the date and company which prepared any such reports as long as such information is not confidential). (Purchaser acknowledges and agrees that Purchaser, its affiliates, agents, employees, representatives, contractors and/or its consultants shall not use or rely on the Environmental Reports except as expressly permitted therein). To Seller's knowledge, except as set forth in the Environmental Reports neither Seller, JPMIM nor ESG has received any written notice from any governmental or quasi-governmental authority of any material violation or liability against the Real Property under Environmental Laws.
- (y) To Seller's knowledge, Schedule "8" annexed hereto and made a part hereof sets forth all insurance policies maintained by the Borrower for the benefit of Seller covering the Real Property, the limits of coverage, deductible amounts and expiration dates of such policies, and such insurance policies are in full force and effect.
- (z) To Seller's knowledge, except as set forth in the Loan Documents or the Ancillary Documents, neither Seller, JPMIM nor ESG has made any written arrangements on behalf of Seller with the Borrower or any person listed on Schedule "9" annexed hereto and made a part hereof (the "Borrower Affiliates") to pay the Borrower or any Borrower Affiliates any fees, commissions or other sums, which have not been paid.
- (aa) To Seller's knowledge, (i) a true, correct and complete copy of the Rosman Ground Lease (as defined in Schedule "10" annexed hereto and made a part hereof) has been made available to Purchaser, (ii) the Rosman Ground Lease is in full force and effect, and (iii) no written notice has been sent to

Seller or the Borrower claiming that the Borrower is in default under the Rosman Ground Lease, which default remains uncured. Seller shall request that the landlord under the Rosman Ground Lease execute an estoppel certificate in the form required under the Rosman Ground Lease (a "Rosman Estoppel"), it being understood and agreed that (A) Seller shall use reasonable efforts to obtain the Rosman Estoppel (provided, however, that Seller shall have no obligation to commence any action or proceeding or to expend any amounts (other than de minimis amounts) to obtain a Rosman Estoppel), (B) the execution and/or delivery of a Rosman Estoppel shall not be a condition to the Closing, (C) if as of the Closing Date, Seller is unable to obtain a Rosman Estoppel, dated within 37 days of the Scheduled Closing Date, as adjourned by Seller, which is not inconsistent, in any material respect, with the representations and warranties set forth in subclauses (i) (ii) and (iii) of this clause (aa) (a "Qualified Rosman Estoppel"), then, in such event, (x) Seller shall have the right, at its option, to adjourn the Closing for up to 45 days in the aggregate (within the 45-day period referred to in section 6.2(f) below) to obtain a Rosman Estoppel, and (y) if Seller has not obtained a Qualified Rosman Estoppel prior to the Closing, as may be adjourned, Purchaser and Seller shall consummate the transaction contemplated hereby subject to and in accordance with the terms of this Agreement, including, without limitation Seller's obligation to make the delivery in Section 6.2(o) hereof; [redacted].

(bb) To Seller's knowledge, (i) [redacted] is the owner of the Share (as hereinafter defined) by assignment from [redacted], and (ii) annexed hereto and made a part hereof as Exhibit "H-2" is a true, correct and complete copy of (x) the Certificate of Amendment of the Certificate of Incorporation of 919T Corp. relating to the Share, (y) the Subscription Agreement relating to the Share executed by [redacted], and (z) the By-Laws of 919T Corp. as in effect on September 21, 1993, and to Seller's knowledge, such Subscription Agreement and By-Laws have not been amended, modified or supplemented in any material respect.

As used in this Agreement, the words "Seller's knowledge" or words of similar import shall be deemed to mean, and shall be limited to, the actual (as distinguished from implied, imputed or constructive) knowledge of [redacted], individuals who have been charged with management responsibility for Seller with respect to the Loan, the Loan Documents and the Real Property without such person having any obligation to make an independent inquiry or investigation, it being understood and agreed that neither individual shall have any liability or obligation hereunder whatsoever.

- 5.2. Breaches of Representations and Warranties. Notwithstanding anything contained herein to the contrary, if the Closing occurs, any claim that Purchaser may have that there has been a breach of any representation or warranty contained herein shall be deemed forever waived by Purchaser (and shall be deemed waived for the purposes of Section 2.1(a) hereof and Section 5.4 hereof) if such breach is known to Purchaser on or before the Closing and an allegation of such breach is not given to Seller in writing on or prior to the Closing Date. For purposes of this Agreement, a breach shall be "known" to Purchaser only if Richard Conniff, Todd Rechler, Gregg Rechler, Jason Barnett, Tod Waterman (collectively, the "Knowledge Persons" and individually, a "Knowledge Person") shall have (x) actual knowledge of such breach, and/or (y) actual knowledge of any information which is the basis of such breach; it being understood and agreed that supplementing the foregoing, each of Knowledge Person shall be deemed to have actual knowledge of (A) all of the documents, instruments, agreements or other materials set forth on Schedule "10" hereto (collectively and, subject to the terms of Section 11.20 hereof, as the same may be amended, modified or supplemented from time to time prior to the Closing, the "Review Materials" and all information contained therein) to the extent held as contemplated by the terms of Section 11.20 hereof, and (B) all written reports or other written materials prepared by Fried, Frank, Harris, Shriver & Jacobson LLP or Ernst & Young LLP and furnished to Purchaser or its affiliates relating to the Loan, the Real Property and/or the transactions contemplated hereby, and all information contained therein. The parties hereto acknowledge and agree that no Knowledge Person shall have any liability hereunder whatsoever.
- 5.3. Covenants of Seller as to the Real Property and Loan. (a) Except as set forth on Schedule "6" annexed hereto and made a part hereof (which sets forth actions Seller shall be entitled to take notwithstanding any limitations set forth in this Agreement) or as may be expressly permitted hereby, so long as Purchaser is not in default of any of the terms, covenants or conditions on Purchaser's part to be observed or performed, Seller shall not, at any time after the date hereof and prior to the Closing, without Purchaser's prior written consent, which consent, in the case of clauses (i), (iii), (iv), (vi) and (viii) below, shall not be unreasonably withheld, delayed or conditioned, (i) consent to the Borrower entering into any new service contracts which cannot be cancelled, without cost to the Borrower (other than de minimis amounts) on not more than thirty (30) days notice or construction agreements, (ii) release the Borrower or any portion of the Real Property from the lien of the Mortgage, (iii) consent to the Borrower amending, modifying, terminating or supplementing any Lease, (iv) consent to the Borrower amending, modifying or supplementing any Service Agreement or Construction Agreement in any material respect, (v) forgive, waive, reduce or compromise any of the indebtedness evidenced or secured by the Loan Documents, nor grant any waiver with respect to any obligation thereunder or under the Ancillary Documents, (vi) modify,

amend or supplement the Approved Budget, (vii) terminate, cancel, modify amend, rescind or supplement the Loan Documents or the Ancillary Documents, or (viii) consent to the Borrower entering into any new lease of any portion of the Real Property; provided, however, that notwithstanding the foregoing to the contrary, Seller's covenants set forth in Section 5.3 hereof shall terminate and shall be of no further force or effect as of the Covenant Termination Date (as hereinafter defined), but following the Covenant Termination Date, Seller shall continue to service and deal with the Loan, the Real Property and the Borrower in accordance with the Approved Budget and otherwise in a reasonably prudent manner in accordance with past practices, it being understood and agreed that supplementing the foregoing proviso, following the Covenant Termination Date and prior to the Closing, Seller (or JPMIM on Seller's or the Borrower's behalf) shall have the right (but not the obligation) to take any action (including, without limitation, make any expenditure or modify the Approved Budget) or refrain from taking any action in the case of an emergency or which, in Seller's reasonable judgment, either would increase the value of the Loan and/or the Real Property. For the purposes of this Agreement, the term "Covenant Termination Date" shall be (I) with respect to the covenants in clauses (i), (iii), (iv), (vi) and (viii) above, 60 days, and (II) with respect to the covenants in clauses (ii), (v) and (vii) above, 120 days, in either case, following the later to occur of (x) the Scheduled Closing Date (as hereinafter defined), or (y) such later date as Seller shall have adjourned the Closing pursuant to the terms of this Agreement.

- (b) Between the date hereof and the Closing Date, so long as Purchaser is not in default of any of the terms, covenants or conditions on Purchaser's part to be observed or performed, Seller shall promptly deliver to Purchaser copies of all material notices given or received by Seller with respect to the Loan and the Real Property, including, without limitation, any notice from the Borrower (and Seller shall request, within two (2) business days following the date hereof, that JPMIM and ESG forward to Seller any such notices received by them).
- (c) Subject to the terms of Section 5.3(a) hereof, between the date hereof and the Closing Date, Seller shall continue to service and deal with the Loan and the Borrower in accordance with the Approved Budget and otherwise in accordance with its prior practices.
- (d) Seller represents to Purchaser that Seller has been advised by JPMIM and ESG that each such party shall be available to be retained by Purchaser in their respective capacities as asset manager and managing and leasing agent for a period of at least thirty days following the Closing, subject to the reasonable negotiation of market terms for providing such services.
- (e) Seller shall request that the Borrower execute and deliver on or before the Closing an estoppel certificate in the form annexed hereto and made a part hereof as Exhibit "B", it being understood and agreed that the execution and delivery by Borrower of an estoppel certificate shall not constitute or be deemed to constitute a condition to Purchaser's obligations hereunder.
- (f) Without limiting the generality of the foregoing, until the Closing, Seller shall cause the insurance policies and the insurance coverages set forth on Schedule "8" hereto (or policies and coverages substantially similar thereto) to remain in full force and effect.
- 5.4. Survival of Representations and Warranties; Holdback. The representations, warranties and covenants of Seller set forth in Section 5.1 hereof shall not merge into any instrument of assignment or transfer delivered at the Closing, and shall survive the Closing for a period of [redacted] (the "Survival Expiration Date"); provided, however, that notwithstanding anything contained herein to the contrary, (a) any allegation of a breach of any representation, warranty or covenant shall be subject to and limited by the terms of Section 5.2 hereof, (b) subject to and limited by the terms of this Section 5.4 and Section 5.2 hereof, Purchaser shall not be entitled to obtain any recovery on account of any breach of any representation, warranty or covenant until the aggregate losses (excluding consequential damages but including any net diminution in the value of the Loan (or, after Closing, the Real Property if Purchaser or its Affiliate (as hereinafter defined) acquires the Real Property)) incurred by Purchaser on account of all breaches of representations and warranties by Seller under Section 5.1 hereof exceeds an amount (the "Representations Threshold Amount") equal to [redacted], less any amounts expended by Purchaser from its own funds pursuant to the terms of clause (II) in Section 5.1(aa) hereof, and (c) subject to and limited by the terms of Section 5.4 and Section 5.2 hereof, Seller's liability at any time with respect to any breach of any representation or warranty shall be limited to the amount of the Closing Escrow Fund (as hereinafter defined) then being held by Escrow Agent. At the Closing, a portion of the Purchase Price, in the amount of [redacted], shall be retained by the Escrow Agent (as such amount may be reduced from time to time, the "Closing Escrow Fund"), to be held by Escrow Agent in escrow subject to and in accordance with the terms of this Section 5.4 and subject to the terms of Section 1.3(f) hereof. Escrow Agent shall hold the Closing Escrow Fund in an interest-bearing account (or as otherwise agreed in writing by Seller, Purchaser and Escrow Agent) in a New York Clearing House Bank or in a nationally recognized "money-market fund." If prior to the [redacted] of the Closing Date, Purchaser has not given to Seller and Escrow

Agent written notice of losses (excluding consequential damages but including any net diminution in the value of the Loan (or, after the Closing, the Real Property if Purchaser or its Affiliate acquires the Real Property)) incurred by Purchaser in excess of the Representations Threshold Amount on account of breaches by Seller of representations and warranties under Section 5.1 hereof or any of the other covenants and agreements of Seller under this Agreement which survive the Closing, then, in such event, on the [redacted] of the Closing Date, Escrow Agent shall pay to Seller (without any notice to Purchaser being required) the amount of [redacted]. If prior to the Survival Expiration Date, Purchaser has not given to Seller and Escrow Agent written notice of losses (excluding consequential damages but including any net diminution in value of the Loan (or, after Closing, the Real Property if Purchaser (or its Affiliate acquires the Real Property)) incurred by Purchaser in excess of the Representations Threshold Amount on account of breaches by Seller of representations and warranties under Section 5.1 hereof or any of the other covenants and agreements of Seller under this Agreement which survive the Closing, then, insuch event, on the Survival Expiration Date, Escrow Agent shall pay to Seller (without any notice to Purchaser being required) the balance of the Closing Escrow Fund. If at any time prior to the Survival Expiration Date, (a) Purchaser has given to Seller and Escrow Agent written notice (which notice Escrow Agent shall confirm receipt of by written notice to Purchaser given within three (3) business days of receipt) of losses excluding consequential damages but including any net diminution in the value of the Loan (or, after Closing, the Real Property if Purchaser or its Affiliate acquires the Real Property) incurred by Purchaser in excess of the Representations Threshold Amount on account of breaches by Seller of representations and warranties under Section 5.1 hereof or any of the other covenants and agreements of Seller under this Agreement which survive the Closing (which notice from Purchaser shall include in reasonable detail the basis for Purchaser's allegation and the calculation of Purchaser's losses), and (b) within thirty-days after such notice, Purchaser shall have commenced an action in a court of competent jurisdiction asserting a claim under this Section 5.4, then, in such event, (x) if the Representations Threshold Amount has been exceeded, with respect to each such claim by Purchaser, Escrow Agent shall retain an amount (the "Escrow Retention Amount") equal to 120% of the lesser of (A) the amount set forth in Purchaser's notice, or (B) the amount of Purchaser's claim in Purchaser's action described in clause (b) of this sentence, until the receipt of joint written instructions from Seller and Purchaser directing the release of the Escrow Retention Amount or Escrow Agent is otherwise directed by a court of competent jurisdiction to release the Escrow Retention Amount, and (y) Escrow Agent shall pay to Seller the remaining Closing Escrow Fund in excess of the Escrow Retention Amount in accordance with, and at the times set forth in, the immediately preceding sentences as if Purchaser had not given to Seller and Escrow Agent any written notice under this Section 5.4 (without any notice to Purchaser being required). Escrow Agent shall have the right, at any time, to deposit all or any portion of the Closing Escrow Fund with a court of competent jurisdiction and thereby be relieved and discharged of any further obligations under this Section 5.4. Escrow Agent shall give written notice of any such deposit to Purchaser and Seller. The provisions of this Section 5.4 shall survive the Closing. For the purposes of this Section 5.4, the term "net diminution in value in the Loan" (or, after Closing, the Real Property if Purchaser or its Affiliate acquires the Real Property) shall be based on the facts and information known to Purchaser (as defined in Section 5.2 hereof) as of the Closing compared with the actual facts and information which result in the breach. For the purposes of the foregoing, Seller and Purchaser acknowledge and agree that there is no need to determine the actual value of the Loan (or the Real Property, as the case may be) as of any point in time, but only the net reduction, if any, in value. The Purchaser and Seller acknowledge and agree that the Closing Escrow Fund may be also used for the breach of any covenants of Seller which survive the Closing, or the purposes set forth in Section 5.1(aa), in each case, without regard to the Representations Threshold Amount.

ARTICLE VI

Closing; EXPENSES AND APPORTIONMENTS

- 6.1. Place of Closing. (a) The consummation of the sale and purchase contemplated in this Agreement (the "Closing") shall take place at the offices of Gibson, Dunn & Crutcher LLP 200 Park Avenue, New York, New York 10166 at 10:00 A.M. on the tenth (10th) business day following the Due Diligence Expiration Date (as hereinafter defined), (the "Scheduled Closing Date", the date the Closing actually occurs being the "Closing Date").
- (b) Subject to Seller's right to adjourn the Closing set forth in this Agreement, each party hereto shall have the one-time right, upon written notice to the other no later than two (2) business days prior to the Scheduled Closing Date, to adjourn the Scheduled Closing Date for up to seven (7) days in the aggregate.
- 6.2. Items to Be Executed or Delivered by Seller. At the Closing, Seller shall execute, acknowledge (where appropriate) and deliver, or cause to be executed, acknowledged (where appropriate) and delivered, the following:
 - (a) The Assignment and Assumption of Mortgage and ss. 275 Affidavit in the form of Exhibit "C" annexed hereto and made a part hereof.

- (b) The original executed Note endorsed to Purchaser with the form of endorsement set forth on Exhibit "D" annexed hereto and made a part hereof (at the Closing, the Note also shall be registered in accordance with the terms of Paragraph 17(q) of the Note); together with all original notes listed on Exhibit "D-1" annexed hereto and made a part hereof.
- (c) The Assignment and Assumption of Collateral Agreements in the form of Exhibit "E" annexed hereto and made a part hereof.
- (d) UCC-3 Financing Statements with respect to each UCC-1 Financing Statements listed on Exhibit "A" hereto.
- (e) An updated title report showing any title matters since the date of Lender's Title Policy (it being understood and agreed that Purchaser's satisfaction with such title report shall not constitute or be deemed to constitute a condition to Purchaser's obligations hereunder).
- (f) Promptly after the date hereof, Seller shall deliver to tenants and Seller shall use its reasonable efforts to obtain an executed tenant estoppel letter (a "Tenant Estoppel") from each of the tenants under the Leases in the form which has been separately approved in writing by the parties hereto; provided, however, that Seller shall have no obligation to bring any action or proceeding or to incur any costs or expenses, other than de minimis amounts, in connection therewith. Each Tenant Estoppel shall be dated within thirty-seven (37) days of the Scheduled Closing Date, as adjourned by Seller. Notwithstanding the foregoing to the contrary, Seller's obligation to deliver Tenant Estoppels shall not be a condition to Closing hereunder and Seller's failure to deliver any Tenant Estoppels shall not diminish or otherwise affect Purchaser's obligations hereunder, except that Seller's obligation to obtain Tenant Estoppels from the tenants listed on Exhibit "F" annexed hereto and made a part hereof shall be a condition to Purchaser's obligations hereunder; provided, however, that in the event that Seller is unable to obtain such Tenant Estoppels, then, in such event, Seller shall have the right, at its sole option, to adjourn the Closing for up to 45 days in the aggregate (without duplication of the 45 day period referred to in Section 5.11(aa)) to obtain such Tenant Estoppels. If the Seller shall have failed to obtain the required Tenant Estoppels which are a condition to the Closing hereunder at or prior to Closing (as the same may be adjourned by Seller hereunder), then, in such event, Purchaser, in its sole discretion, shall have the right, exercised no later than the Scheduled Closing Date (as the same may be adjourned by Seller or Purchaser as permitted hereunder), either to (i) waive such condition with respect to Tenant Estoppels, accept the Loan and consummate the Closing without abatement of, or credit against, the Purchase Price and without liability on the part of Seller, or (ii) exercise the Termination Option. To the extent that (i) any Tenant Estoppel obtained from a tenant contains information which has been represented by Seller in Seller's representations in Section 5.1 hereof and such Tenant Estoppel is contrary to, in any material respects, such representations, then, in such event, Seller shall be deemed to have breached such representations but Seller shall be deemed to have satisfied all of the conditions with respect to such Tenant Estoppel set forth in this Section 6.2(f) notwithstanding such disagreement, or (ii) any Tenant Estoppel obtained from the tenant does not contain information described in the preceding subclause (i) and otherwise disagrees with any information set forth in such Tenant Estoppel in the form sent to such tenant (or otherwise sets forth additional information) Seller shall be deemed to have satisfied all of the conditions with respect to such Tenant Estoppel as set forth in this section 6.2(f) notwithstanding such facts or information.
- (g) A closing statement reflecting all of the financial aspects of the transaction (the "Closing Statement") initialed by Seller.
- (h) A non-foreign status affidavit in the form annexed hereto and made a part hereof as Exhibit "G".
- (i) The Assignment and Assumption of Share in the form of Exhibit "H" annexed hereto and made a part hereof with respect to the share (the "Share") of 919T Corp. held by Takeshi Saeki, together with (A) the original Share, and (B) an executed stock power in the form of Exhibit "H-1" hereto.
- (j) Any and all documents and/or instruments reasonably necessary to change (i) the name of any Cash Collateral Account from the name of Seller to Purchaser, and (ii) the signatures of any Cash Collateral Account.
- (k) An original of each Loan Document to the extent in Seller's possession.
- (1) A copy of each Lease (together with a copy of such information related to the Leases which is reasonably necessary for an owner of the Real Property to administer the Leases) to the extent in Seller's possession.

- (m) A certificate, dated as of the Closing Date, stating that the representations and warranties of Seller set forth in Section 5.1 hereof are true and correct in all material respects as of the Closing Date (with appropriate modifications of such representations and warranties to reflect any changes therein, including, without limitation, any changes resulting from actions under Section 5.3 hereof) or identifying any representation or warranty which is not, or no longer is, true and correct in all material respects.
- (n) Either (i) an opinion of Seller's outside counsel as to due authorization, execution and delivery in customary form and subject to customary exceptions, assumptions and qualifications, or (ii) if Purchaser's counsel and Seller's counsel are unable to agree on the form of the opinion referred to in clause (i) of this sentence, then, in such event, (I) a certificate ("Seller's Authority Certificate") from Seller, executed by the general partner in Seller and the limited partners in Seller owning more than a majority-in-interest in Seller, consenting, without condition, to the transactions contemplated by this Agreement, together with (x) a certified copy of Seller's existing agreement of limited partnership, which agreement of limited partnership shall be in such a form that the consents, as provided herein, are sufficient to authorize Seller to enter into the transactions contemplated by this Agreement and consummate the transactions contemplated by this Agreement, and (y) a legal opinion from Japanese counsel (or counsel from such other jurisdiction which the limited partners are organized under, if not Japan) as to due authorization, execution and delivery of Seller's Authority Certificate by any limited partners in customary form and subject to customary exceptions, assumptions and qualifications; and (II) a Secretary's Certificate from an officer of the general partner in Seller certifying to the adoption and non-revocation of resolutions duly authorizing the transactions contemplated by this Agreement, together with a certified copy of the general partner's certificate of incorporation and bylaws and an incumbency certificate for the natural person signing on behalf of the general partner.
- (o) If Seller is unable to obtain a Qualified Rosman Estoppel as provided in Section 5.1(aa) hereof, then, in such event, the letter in the form annexed hereto and made a part hereof as Schedule "12".
- (p) All other instruments and documents which may be reasonably required to effect the transaction contemplated by this Agreement and within Seller's control, provided that such instruments or documents may be delivered without additional cost or liability to Seller (other than de minimis amounts).
- $6.3.\ \mbox{Items}$ to Be Delivered or Executed by Purchaser. At the Closing, Purchaser shall:
 - (a) Pay to Seller the Purchase Price, subject to the apportionments and adjustments set forth herein, in the manner set forth in Section 1.2 hereof.
 - (b) Execute, acknowledge (where appropriate) and deliver, or cause to be executed, acknowledged (where appropriate) and delivered, the following:
 - (i) The Assignment and Assumption of Mortgage in the form of Exhibit "D" hereto.
 - (ii) The Assignment and Assumption of Collateral Agreement in the form of Exhibit $\mbox{"E"}$ hereto.
 - (iii) The Assignment and Assumption of Share in the form of Exhibit "H" hereto.
 - (iv) Any and all documents and/or instruments reasonably necessary to change (A) the name of any Cash Collateral Account from the name of Seller to Purchaser, and (B) the signatures of any Cash Collateral Account.
 - (v) The Closing Statement initialed by Purchaser.
 - (vi) All other instruments and documents which may be reasonably requested to effect the transaction contemplated by this Agreement and within Purchaser's control, provided that such instruments or documents may be delivered without additional cost or liability to Purchaser (other than de minimis amounts).
 - (c) An opinion of Purchaser's outside counsel as to due authorization, execution and delivery in customary form and subject to customary exceptions, assumptions and qualifications.
- 6.4. Taxes; Title Expenses. (a) Any transfer, documentary, stamp, gains and similar taxes and any filing fees (collectively, "Taxes") imposed in connection with the transfer of the Loan or recording any document or instrument contemplated hereby shall be paid by Purchaser, and all reports,

returns and related materials required to be submitted in connection therewith shall be submitted by Seller at or prior to the Closing and evidence of the same shall be submitted to Purchaser at the Closing.

- (b) All premiums and fees or other costs for title examination and title insurance or associated with any update of or endorsement to the Lender's Title Policies, or any other report, study, survey or diligence research obtained by Purchaser, if any, and all related charges in connection therewith shall be paid by Purchaser.
- 6.5. Apportionments. (a) At the Closing, Purchaser shall pay to Seller an amount (the "Collateral Account Closing Amount"), equal to all of the funds in the Cash Collateral Account as of the Closing Date (as set forth on account statements for such accounts as of the Closing Date), which amount shall not exceed \$18,000,000, it being understood and agreed that (i) there shall be no prorations with respect to the Collateral Account Closing Amount, and (ii) all credits and debits determined as a result of any apportionments and adjustments under this Agreement shall only be to or against the Purchase Price.
- (b) To the extent, on or before the Closing Date, Seller has not paid or caused to be paid by the Borrower (or any other person) the amounts set forth on Schedule "7" annexed hereto and made a part hereof on account of capital expenditures, leasing expenses and commissions and tenant improvement work related to the Real Property for which Schedule "7" hereto states Seller is responsible, Purchaser shall receive a credit against the Purchase Price. The failure to make or cause to be made the payments described in the immediately preceding sentence shall not be deemed to result in a "net diminution in value of the Loan" for purposes of Section 8.3 hereof (because Purchaser shall receive a credit against the Purchase Price for such unpaid item as set forth in the preceding sentence) except to the extent the delay in payment as opposed to non-payment itself has resulted in a "net diminution in value of the Loan" for the purposes of Section 8.3. Nothing contained in the preceding sentence shall be construed to effect a "net diminution in value of the Loan" for the purpsoes of Section 8.3 arising out of the status of the work paid or caused to be paid for by the Seller or credited to Purchaser as set forth in this Section 6.5(b). With respect to the items of capital expenditures, leasing expenses and commissions, tenant have resulted in a "net diminution in value of the Loan" for the purposes of Section 8.3. With respect to the items of capital expenditures, leasing expenses and commissions, tenant improvement work related to the Real Property described on Schedule "7" hereto, Seller shall not have any further requirements to make any payments. Nothing in this clause (b) shall be in derogation of Purchaser's rights with respect to breaches of representations set forth in Section 5.1 hereof (after taking into account the apportionments made or to be made under this Section 6.5(b)).
- (c) All operating expenses with respect to the Real Property will be apportioned between the parties on and as of the Closing Date based upon documentary evidence of payments made or due for such operating expenses.
- (d) (i) Except as set forth in clause (ii) below, minimum rent, additional rent, expense escalations and other adjustments and charges under, or in respect of, the Leases shall be apportioned as of the Closing Date as, when and to the extent actually collected.
 - (ii) If, as of the Closing Date, any tenant or occupant of the Real Property shall be in arrears in the payment of minimum or additional rent, expense escalations or other adjustments, then, in such event payments received from such tenants or occupants after the Closing shall be paid and applied in the following order of priority: (I) first, apportioned between Purchaser and Seller on account of the month in which the Closing occurred; (II) second, to Seller on account of the month preceding the Closing; (III) third, to Purchaser on account of any month or months following the Closing; and (IV) fourth, to Seller on account of any month or months preceding the month in which the Closing occurred;
 - (iii) After the Closing, Purchaser shall (I) use its reasonable efforts to collect, on behalf of Seller, all accrued minimum and additional rent, expense escalations, tax and other adjustments, and other charges in which Seller has an interest which have not been collected prior thereto, and (II) subject to clause (ii) above, deliver to Seller all such amounts collected (less any reasonable expenses incurred by Purchaser in collecting such amounts), together with a reasonably detailed accounting of the amounts paid by tenants and the amounts due to Seller from Purchaser. Notwithstanding the foregoing to the contrary (x)Purchaser shall have no obligation to commence (and Seller shall not commence) any action or proceeding to collect any arrearages which may be due Seller, and (y) Purchaser's obligations under clause (I) shall expire on the first anniversary of the Closing Date with respect to arrearages of minimum rent, and shall expire on the first anniversary of the date after the Closing Date on which arrearages of additional rent, expense escalations, tax or other adjustments are determined.
 - (iv) (A) If, prior to the Closing, Seller shall receive any installments of additional rent, expense escalations, tax or other adjustments from any tenant attributable to periods after the Closing Date, then, Purchaser shall receive a credit against the Purchase Price in

the amount of such installments. In addition, to the extent that tenants' additional rent, expense escalations, tax or other adjustments for the period prior to the Closing have been paid by tenants based on estimates, Seller shall promptly refund to the applicable tenant any overpayments of additional rent, expense escalations, tax or other adjustments which is determined to be owed to such tenant subsequent to the Closing, it being understood and agreed that if Seller shall fail to promptly pay any amount determined to be so owed, then, in such event, upon written notice to Seller, Purchaser may pay such amounts directly to the applicable tenant and obtain reimbursement therefor from the Closing Escrow Fund. The provisions of this clause (A) shall survive the Closing.

- (B) To the extent that tenants' additional rent, expense escalations and/or tax or other adjustments for the period prior to the Closing have been paid by tenants based on estimates, subject to the terms of clause (ii) above, Purchaser shall promptly pay to Seller any underpayments of additional rent, expense escalations and/or tax or other adjustments which are actually paid by the tenants to Purchaser and determined to be owed to Seller subsequent to the Closing. The terms of this clause (B) shall survive the Closing.
- (e) (i) Real estate taxes shall be apportioned between Seller and Purchaser at the Closing on the basis of the fiscal year for which assessed. If the Closing shall occur before a new tax rate is fixed, the apportionment of real estate taxes at the Closing shall be based upon the tax rate for the immediately preceding fiscal period applied to the latest assessed valuation of the Real Property. Promptly after the new tax rate has been fixed, the apportionment of real estate taxes made at the Closing shall be recomputed. Any installment of any assessment levied against the Real Property which is due prior to the Closing Date shall be paid in full by Seller or the Borrower, and Purchaser shall be responsible for any remaining installments which are due and payable after the Closing Date.
- (ii) Any real estate tax refunds or credits with respect to the Real Property which are attributable to the fiscal period in which the Closing occurs shall be apportioned between Seller and Purchaser based upon the time period in which the refunds or credits relate, after deducting all costs and expenses of collecting same. Any real estate tax refunds or credits with respect to the Real Property which are received after the Closing Date and which are attributable to any fiscal period prior to the fiscal period in which the Closing occurs shall belong solely to Seller. Seller, or its agents, on behalf of the Borrower may withdraw, settle or otherwise compromise any protest or reduction proceeding effecting real estate taxes assessed against the Real Property for any fiscal period prior to the Closing with Purchaser's consent, which consent shall not be unreasonably withheld, delayed or conditioned. Neither Purchaser nor its agents, on behalf of the Borrower or otherwise, shall withdraw, settle, or otherwise compromise any protest or reduction proceeding effecting real estate taxes assessed against the Real Property for the fiscal period in which the Closing occurs without the prior written consent of Seller, which consent shall not be unreasonably withheld, delayed or conditioned. With respect to real estate tax refunds or credits with respect to the Real Property relating to the fiscal year in which the Closing occurs or relating to prior fiscal years which may be required to be reimbursed to tenants at the Real Property, Purchaser and Seller shall enter into an escrow arrangement on or prior to the Due Diligence Expiration Date, in form and substance reasonably satisfactory to Purchaser and Seller, for the payment of such reimbursements (it being understood and agreed that the parties hereto hereby approve Pottish Freyberg Marcus & Velazquez, LLP as escrow agent for the purposes of such escrow arrangement). The terms of this Section 6.5(d) shall survive the Closina.
- (f) All insurance policies currently maintained by the Borrower covering the Real Property shall be terminated on the Closing Date, consequently there shall be no apportionment of any insurance premiums.
- (g) (I) If after the date hereof, the Borrower enters into any Lease or if there is any extension or renewal of any Leases, whether or not such Leases provide for their extension or renewal, or any expansion or modification of any Leases (each, a "New Lease"), in each case to the extent entered into in accordance with this Agreement, Seller shall keep accurate records of all expenses (collectively, "New Lease Expenses") incurred in connection with each New Lease, including, without limitation, the following: (i) brokerage commissions and fees relating to such leasing transaction, (ii) expenses incurred for repairs, improvements, equipment, painting, decorating, partitioning and other items to satisfy the tenant's requirements with regard to such leasing transaction, (iii) the cost of removal and/or abatement of asbestos or other hazardous or toxic substances located in the demised space, (iv) reimbursements to the tenant for the cost of any of the items described in the preceding clauses (ii) and (iii), (v) legal fees for services in connection with the preparation of documents and other services rendered in connection with the effectuation of the leasing transaction, (vi) rent concessions relating to the demised space provided the tenant has the right to take possession of such demised space during the period of such rent concessions, and (vii) expenses incurred for the purpose of satisfying or terminating the obligations of a tenant under a New Lease to the landlord under another lease (whether or not such other lease covers space in the Real Property).

(II) The New Lease Expenses for each New Lease allocable to Seller shall be determined by multiplying the amount of such New Lease Expenses by a fraction, the numerator of which shall be the number of days contained in that portion, if any, of the term of such New Lease commencing on the date on which the tenant thereunder shall commenced to pay fixed rent ("Commencement Date") and expiring on the date immediately preceding the Closing Date, and the denominator of which shall be the total number of days contained in the period commencing on the Commencement Date and expiring on the date of the scheduled expiration of the term of such New Lease, and the remaining balance of the New Lease Expenses for each New Lease shall be allocable to and the responsibility of Purchaser. For purposes of this clause (II), the Commencement Date under a renewal, extension, expansion or modification of a Lease shall be deemed to be (i) in the case of a renewal or extension (whether effective prior to or after the Closing, or in the form of an option exercisable in the future), the first date during such renewal or extension period after the originally scheduled expiration of the term of such Lease on which the tenant under such Lease commences to pay fixed rent, (ii) in the case of an expansion (whether effective prior to or after the Closing, or in the form of an option exercisable in the future), the date on which the tenant under such Lease commences to pay fixed rent for the additional space, and (iii) in the case of a modification not also involving a renewal, extension or expansion of such Lease, the effective date of such modification agreement. At the Closing, Purchaser shall reimburse Seller for all New Lease Expenses theretofore paid by Seller or the Borrower, if any, in excess of the portion of the New Lease Expenses allocated to Seller pursuant to the provisions of the preceding sentence.

(h) All apportionments shall be computed as of the close of business on the day immediately preceding the Closing Date and such apportionments shall be calculated in accordance with the "customs in respect of title closings" of the Real Estate Board of New York on the Closing Date. Any calculation made in computing any apportionment made pursuant to this Section 6.5 may be reconfirmed following the Closing, and any errors thereto shall be corrected immediately upon notice from the other party that such error(s) exist. If it is determined that Seller failed to cause the Borrower to pay any expense with respect to the Real Property not apportioned hereunder for any period prior to the Closing Date, which expense was incurred through the act, with the consent or at the direction of JPMIM, then, Seller shall pay or cause to be paid such expense; it being understood and agreed that (i) the terms of this sentence shall not apply to any matter described on Schedule "7" hereto, (ii) Seller shall be entitled to use the Closing Escrow Fund to satisfy its obligations under this sentence, and (iii) Seller's liability under this sentence shall be limited to the then existing amount of Closing Escrow Fund. The provisions of this Section 6.5 shall survive the Closing for six (6) months.

ARTICLE VII

DEFAULTS

- 7.1. Purchaser Default. Provided that Seller is (i) not in material default of its non-monetary obligations hereunder, (ii) not in default, in any respect, of its monetary obligations hereunder, and (iii) is otherwise ready, willing and able to consummate the transactions contemplated hereby, in the event Purchaser shall default in the performance of its obligation to purchase the Loan in breach of its obligations under this Agreement, Seller shall be entitled to terminate this Agreement and the sole remedy of Seller shall be to retain the Escrow Funds as liquidated damages for all loss, damage and expense suffered by Seller on account thereof, it being acknowledged by Purchaser and Seller that in such event Seller will suffer substantial damages but such damages are incapable of exact ascertainment. After payment to Seller of the Escrow Funds, neither Seller nor Purchaser shall have any further rights or obligations hereunder; provided, however, (a) that the indemnities contained in Section 8.2(b) and Section 11.2 hereof, and (ii) the terms of the Confidentiality Agreement, shall survive any such termination. Notwithstanding anything to the contrary herein contained, if subsequent to the Closing Purchaser shall fail to comply with its obligations contained herein which survive Closing, Seller in addition to any rights and remedies provided herein, shall be entitled to any and all remedies available at law, in equity or otherwise.
- 7.2. Seller Default. (a) Provided that Purchaser is (i) not in material default of its non-monetary obligations hereunder, (ii) not in default, in any respect, of its monetary obligations hereunder, and (iii) is otherwise ready, willing and able to consummate the transactions contemplated hereby, in the event that on the Closing Date Seller shall be unable to perform or is prohibited by court order from performing its obligations or to satisfy any condition applicable to Seller hereunder in accordance with the provisions of this Agreement, the sole remedy of Purchaser shall be to exercise the Termination Option.
- (b) Provided that Purchaser is (i) not in material default of its non-monetary obligations hereunder, (ii) not in default, in any respect, of its monetary obligations hereunder, and (iii) is otherwise ready, willing and able to consummate the transactions contemplated hereby, in the event that Seller willfully defaults in its obligation to transfer the Loan hereunder, Purchaser

shall be entitled to such remedies against Seller as are available at law or in equity, including, without limitation, specific performance. Notwithstanding anything to the contrary herein contained, if subsequent to the Closing Seller shall fail to comply with its obligations contained herein which survive Closing, Purchaser in addition to any rights and remedies provided herein, shall be entitled to any and all remedies available at law, in equity or otherwise.

ARTICLE VIII

INSPECTIONS; DUE DILIGENCE

- 8.1. Inspection of the Real Property. At all times prior to Closing or the earlier termination of this Agreement, upon reasonable written notice to Seller, Purchaser and such agents and representatives of Purchaser ("Authorized Representatives") as shall have been identified to JPMIM in writing, shall have the right, subject to the rights of tenants, the Borrower and other occupants at the Real Property, to enter upon the Real Property to inspect and examine the same; provided, however, that (a) Purchaser shall not unreasonably interfere with the Borrower's use or operation of the Real Property, (b) Seller, JPMIM or their representatives or employees shall accompany Purchaser or any Authorized Representatives, (c) Purchaser shall not invade, alter or destroy the Real Property in any manner whatsoever, and (d) as a condition precedent to making any physical inspection of the Real Property, Purchaser shall obtain and maintain (i) liability insurance in the amount of \$2,000,000 for property damage coverage and in the amount of \$5,000,000 for personal injury coverage, which insurance shall name Seller, JPMIM and ESG as additional insured parties, and (ii) with respect to any Authorized Representatives, workers' compensation and disability insurance, as required by law. Purchaser shall provide to JPMIM certificates of insurance evidencing such coverage prior to entering upon the Real Property for the purpose of making any such physical investigation. Purchaser and the Authorized Representatives shall have the right, from time to time, upon reasonable written notice to and during normal business hours, to examine the books, records and accounts with respect to the Real Property at such reasonable location as JPMIM shall designate. It is understood and agreed that Purchaser's satisfaction with such inspection shall not constitute or be deemed to constitute a condition to Purchaser's obligations hereunder.
- 8.2. Purchaser's Work. (a) Any work performed by Purchaser or its agents in connection with any inspection of the Real Property shall be at Purchaser's sole cost and expense. Purchaser covenants and agrees to pay in full all persons who perform labor upon the Real Property, and not to permit or suffer any mechanic's or materialmen's lien of any kind or nature to be asserted or enforced against the Real Property for any work done or materials furnished thereon at the instance or request or on behalf of Purchaser.
- (b) Purchaser shall indemnify, defend and hold harmless Seller, the Borrower, and the Seller Parties from and against any and all losses, costs, liabilities, claims, damages or expenses (including, without limitation, reasonable attorney's fees and costs) arising out of any inspection of, or access to, the Real Property by Purchaser or its Authorized Representatives. Purchaser, at its sole cost and expense, shall promptly restore the Real Property to its condition immediately prior to the performance of such investigation by Purchaser pursuant to this Article VIII and shall repair any and all damage caused by Purchaser or Purchaser's employees, representatives or agents. Purchaser acknowledges and agrees that Seller shall have the right (but not the obligation) to use the Escrow Funds to restore the Real Property if Purchaser shall fail to comply with this Section 8.2. The terms of this Section 8.2 shall survive the Closing or earlier termination of this Agreement.
- 8.3. Due Diligence. (a) Within one (1) day after the date hereof, Seller shall deliver or make available to Purchaser the Review Materials existing as of the date hereof. Subject to and in accordance with the terms of Section 8.1 hereof and Section 8.2 hereof, Purchaser shall have access to the Real Property and information relating to the Real Property.
- (b) Subject to and in accordance with the terms of this Section 8.3, Purchaser shall have the right to exercise the Termination Option if (i) Purchaser's due diligence reveals any fact (a "Diligence Fact") relating to the Loan or the Real Property (but not relating to (x) the business or financial condition of the Borrower (or its direct or indirect partners or their respective affiliates) or the organizational structure of the Borrower (or its direct or indirect partners or their respective affiliates), or (y) the general real estate or financial markets) which was not previously disclosed in Schedule "1" hereto, Schedule "6" hereto or the written information prepared by ESG annexed hereto and made a part hereof as Schedule "11" and/or the PRO-JECT Financial Analysis Disk accompanying such written information (the "ESG Information Book"), and the failure to cure or remediate such Diligence Fact would result in the net diminution in value of the Loan, and (ii) the aggregate amount (the "Diminution Amount") determined under clause (i) of this sentence is in excess of [redacted] Dollars (the "Threshold Amount"). Notwithstanding the foregoing to the contrary but subject to the terms of clause (d) below, if (i) the Diminution Amount, in the aggregate, does not exceed [redacted] Dollars [redacted] (the "Cap Amount"), and (ii) Seller (in its sole discretion and within five (5) business days after Seller's receipt of Purchaser's notice

setting forth in reasonable detail each Diligence Fact and each related Diminution Amount ("Purchaser's Preliminary Notice")) shall agree to a reduction of the Purchase Price by an amount (the "Reduction Amount") equal to the aggregate Diminution Amount in excess of the Threshold Amount, up to the Cap Amount, then, in such event, (A) Purchaser shall be deemed not have the right to terminate this Agreement as provided in this Section 8.3, and (B) the Purchase Price shall be deemed to be equal to the original Purchase Price, reduced by the Reduction Amount as provided herein, for all purposes in this Agreement; provided, however, that Seller shall have the right to extend such 5-business day period for an aggregate of fifteen (15) days. For the purposes of this Section 8.3, the term "net diminution of value" shall mean a net decrease in the value of the Loan resulting from actual facts and information (other than facts and information relating to (x) the business or financial condition of the Borrower (or its direct or indirect partners of their respective affiliates), or (y) the general real estate or financial markets not previously disclosed in Schedule "1", Schedule "6" or the ESG Information Book. For the purposes of the foregoing, Seller and Purchaser acknowledge and agree that there is no need to determine the actual value of the Loan as of any point in time, but only the net reduction, if any, in value.

(c) Purchaser's right to exercise the Termination Option under this Section 8.3 shall be exercisable only by delivery to Seller and Escrow Agent of the Purchaser's Preliminary Notice at any time during the period commencing on the date of this Agreement and continuing up to but no later than 5:00 p.m. on $[redacted] \ (such \ date, \ as \ the \ same \ may \ be \ extended \ pursuant \ to \ this \ Section$ 8.3(c), being the "Due Diligence Expiration Date"); provided, however, that if Seller fails to respond to Purchaser's Preliminary Notice prior to the Due Diligence Expiration Date, Purchaser's right to terminate this Agreement as provided in this Section 8.3 shall be preserved (and the Due Diligence Expiration Date shall be deemed extended) until Purchaser's receipt of Seller's notice stating whether or not Seller agrees to such reduction in the Purchase Price; provided, further, however, that (I) Purchaser shall have no right to deliver any supplement or addition to Purchaser's Preliminary Notice after the Due Diligence Expiration Date and the Due Diligence Expiration Date shall not be deemed extended for that purpose, (II) if Seller notifies Purchaser of its refusal to so reduce the Purchase Price by the Reduction Amount then, in such event, Purchaser shall have the right, by written notice to Seller within one (1) business day following receipt of Seller's notice of its refusal to reduce the Purchase Price, to (i) exercise the Termination Option, or (ii) waive its rights to terminate this Agreement under this Section 8.3 and accept the Loan and consummate the Closing subject to and in accordance with the terms of this Agreement without any abatement of, or credit against, the Purchase Price and Purchaser shall have one (1) additional business day to deliver the Additional Deposit to Escrow Agent, and (III) if Seller notifies Purchaser that Seller agrees to reduce the Purchase Price by the Reduction Amount, then, in such event, Purchaser shall have two (2) business days to deliver the Additional Deposit to Escrow Agent. Notwithstanding anything contained herein to the contrary, (x) if Purchaser fails to deliver a Purchaser's Preliminary Notice by the Due Diligence Expiration Date, then Purchaser shall be deemed to have forever waived its right to terminate this Agreement pursuant to this Section 8.3, it being understood and agreed that Purchaser shall have no right to deliver a Purchaser's Preliminary Notice after the Due Diligence Expiration Date, or (y) if prior to the date which is two (2) business days prior to the Due Diligence Expiration Date Seller gives any of the Knowledge Persons written notice of any fact and/or information which results in a "net diminution value" (as defined in Section 8.3(b) above) and Purchaser does not include any such fact and/or information in Purchaser's Preliminary Notice, then Purchaser shall be deemed to have forever waived its right to assert any rights under this Agreement with respect to any such fact and/or information.

ARTICLE IX

indemnification

9.1. Indemnification. (a) Effective upon the consummation of the Closing, Purchaser shall indemnify, defend and hold harmless Seller and the Seller Parties (collectively, "Indemnitees") from and against any and all Liabilities (as hereinafter defined), which any Indemnitee may suffer, incur or payout, or which may be asserted against any Indemnitee, in whole or in part, by reason of or in connection with, or arising out of the Loan, the Loan Documents or the Real Property (including, without limitation, any claims of third parties, tenants under the Leases (or under any subordination, nondisturbance and attornment agreement entered into in connection with any such Lease), or the Borrower) (i) arising after the Closing Date with respect to events and/or conditions (including, without limitation, acts or omissions) occurring after the Closing Date, or (ii) arising in connection with or out of any enforcement claim, proceeding, suit or other action whatsoever with respect to the Loan taken by Purchaser (or any Affiliate of Purchaser) against the Borrower or its direct or indirect partners or their respective affiliates, it being understood and agreed that in connection with any agreement or other arrangement by and between Purchaser (or its Affiliate) and the Borrower (or its direct or indirect partners or their respective affiliates), Purchaser (or its Affiliate) shall use all commercially reasonable and good faith efforts to obtain a full, unconditional and irrevocable release of Indemnitees with respect to the Loan (and Purchaser (or its Affiliate) upon Seller's (or its representative's) request, shall keep Seller reasonably informed of the status of the negotiation

of any such release, but, without request, Purchaser (or its Affiliate) shall inform Seller if it has obtained such release or if it is unable to obtain such a release). The terms of this Section 9.1 shall survive the Closing.

- (b) For the purposes of this Agreement, the term "Liability" shall mean any liability, obligation, loss, costs, damage, penalty, claim, lien, charge and/or expense (including, without limitation, attorneys' and other professionals' fees and expenses).
- 9.2. Release. Except for Seller's obligations hereunder which survive Closing, effective as of the consummation of the Closing, to the fullest extent permitted by law, Purchaser hereby releases, discharges and forever acquit each Indemnitee from all matters, claims, liabilities, demands, causes of action, debts, obligations, promises, acts, agreements, interests, damages of whatever kind or nature, and any attorneys' fees and expenses that may be related thereto, whether known or unknown which Purchaser has or may hereafter incur or acquire against any Indemnitee by reason of or relating to this Agreement, the Loan, the Loan Documents or the Real Property. The terms of this Section 9.2 shall survive the Closing.

ARTICLE X

casualty and condemnation

- 10.1. Casualty. If, prior to the Closing Date, any damage or destruction (a "Casualty") of any part of the Real Property as a result of any fire or other casualty occurs, Seller shall notify Purchaser of such Casualty. If prior to the Closing Date a Non-Material Casualty (as hereinafter defined) occurs, then, in such event, (a) this Agreement shall remain in full force and effect, and (b) Seller shall (i) credit to Purchaser against the Purchase Price the amount of any insurance proceeds theretofore received by Seller in connection with any such Casualty, less Seller's out-of-pocket cost incurred in connection with its attempts to seek collection of any such insurance proceeds, and credit to Purchaser against the Purchase Price an amount equal to the deductible amount under the Borrower's insurance policies, and (ii) assign to Purchaser by written assignment in form and substances satisfactory to Seller and Purchaser, at the Closing, all of Seller's right, title and interest in and to any insurance proceeds payable in connection with such Casualty. In the event of an occurrence of a Casualty referred to in the immediately preceding sentence prior to the Closing Date, Purchaser shall have the right to participate, in a reasonable manner, in the negotiations and settlement of any insurance claim and any decision regarding the reconstruction or renovation of any improvement on the Real Property, it being understood and agreed that Seller shall not settle any insurance claim or consent to any application of any insurance proceeds to the restoration of the Real Property without Purchaser's prior written consent, which consent shall not be unreasonably withheld, delayed or conditioned. If prior to the Closing Date, a Material Casualty (as hereinafter defined) occurs, Purchaser shall have the right, without duplication of its rights under Section 8.3 hereof, to exercise the Termination Option upon notice to Seller given upon the earlier to occur of (x) ten (10) days after written notice is given to Purchaser of the Casualty, or (y) the Closing Date. If Purchaser elects not to exercise the Termination Option, the Material Casualty shall be deemed to be a "Non-Material Casualty" for all purposes hereunder and Purchaser and Seller shall proceed to Closing, subject to satisfaction of the provisions herein with respect to a Non-Material Casualty. Except as otherwise provided in this Section 10.1, Purchaser shall not be entitled to any reduction of, or credit against, the Purchase Price in the event of any Casualty, and Purchaser shall be obligated to close hereunder notwithstanding the occurrence of any such event. Seller shall in no event have any obligation to Purchaser to repair, restore or replace any portion of the Real Property affected by any Casualty. As used herein, the term "Material Casualty" shall mean any Casualty (i) the cost of which to repair is in excess of [redacted], or (ii) which will allow the tenant under the [redacted] to terminate their Leases. As used herein, the term "Non-Material Casualty" shall mean a Casualty which is not a Material Casualty.
- 10.2. Condemnation. If, prior to the Closing Date, any taking by condemnation or eminent domain (a "Condemnation") of any part of the Real Property occurs, Seller shall notify Purchaser of such Condemnation. If, prior to the Closing Date, a Condemnation occurs, then, in such event, (a) this Agreement shall remain in full force and effect, and (b) Seller shall (i) credit to Purchaser against the Purchase Price the amount of any award theretofore received by Seller in connection with any such Condemnation, less Seller's out-of-pocket cost incurred in connection with its attempts to seek collection of any such awards, and (ii) assign to Purchaser by written assignment in form and substances satisfactory to Seller and Purchaser, at the Closing all of Seller's right, title and interest in and to any award payable in connection with such Condemnation. Prior to the Closing Date, Purchaser shall have the right to participate, in a reasonable manner, in the negotiations and settlement of any Condemnation award and any decision regarding the reconstruction or renovation of any improvement on the Real Property, it being understood and agreed that Seller shall not settle any Condemnation award or consent to any application of any Condemnation award to the restoration of the Real Property without Purchaser's prior written consent, which consent shall not be unreasonably withheld, delayed or conditioned. Except as otherwise provided in this Section 10.2, Purchaser shall not be

entitled to any reduction of, or credit against, the Purchase Price in the event of any Condemnation, and Purchaser shall be obligated to close hereunder notwithstanding the occurrence of any such event. Seller shall in no event have any obligation to Purchaser to repair, restore or replace any portion of the Real Property affected by any Condemnation.

ARTICLE XI

Miscellaneous

- 11.1. No Third-Party Beneficiaries. Nothing in this Agreement is intended to confer upon any person, other than the parties hereto and their respective successors and permitted assigns, any rights, remedies, obligations or liabilities under, or by reason of, this Agreement.
- 11.2 Broker. (a) Each of Seller and Purchaser represents and warrants to the other that it knows of no broker or finder with whom it has dealt or who has claimed or who may have the right to claim any fee, commission or other similar compensation in connection with the transaction contemplated by this Agreement other than ESG (the "Broker"). Seller represents and warrants that it shall be solely responsible for the payment of any and all fees, commissions or other compensation due to the Broker in connection with this transaction pursuant to a separate agreement previously entered into between Seller and Broker.
- (b) Purchaser shall indemnify, defend and hold harmless Seller, its agents, employees and representatives from and against any and all losses, costs, liabilities, claims, damages or expenses (including, without limitation, reasonable attorneys' fees and costs) arising out of the breach of Purchaser's representations or warranties contained in this Section 11.2.
- (c) Seller shall indemnify, defend and hold harmless Purchaser, its agents, employees and representatives from and against any and all losses, costs, liabilities, claims, damages or expenses (including, without limitation, reasonable attorneys' fees and costs) arising out of the breach of Seller's representations or warranties contained in this Section 11.2.
- (d) The representations, warranties and indemnities contained in this Section 11.2 shall survive the Closing or, if the Closing does not occur, the termination of this Agreement.

11.3 [Intentionally Blank.]

- 11.4. Assignment. This Agreement shall be binding upon and inure to the benefit of the parties hereto and their respective heirs, representatives, successors and permitted assigns; provided, however, that Purchaser shall not sell, assign, transfer or otherwise convey its rights or obligations under this Agreement, in whole or part, without the prior written consent of Seller; except, however, that at the Closing, Purchaser shall be entitled to assign its rights under this Agreement to an Affiliate (as hereinafter defined) without Seller's prior written consent provided that (A) Purchaser and such Affiliate execute and deliver to Seller an originally executed Assignment and Assumption Agreement, in form and substance reasonably satisfactory to Seller, pursuant to which such Affiliate assumes all of Purchaser's obligations hereunder, and (B) Purchaser shall remain fully and primarily liable for all obligations and liabilities hereunder (including, without limitation, those that survive the Closing or earlier termination of this Agreement). For the purposes of this Section 11.4, the term "Affiliate" shall mean an entity (x) directly or indirectly controlling, controlled by, or under common control with Purchaser, or (y) fifty percent (50%) or more of whose equity interest is directly or indirectly owned or held by Purchaser.
- 11.5. Waiver of Jury Trial. EACH OF THE PURCHASER AND SELLER HEREBY WAIVES TRIAL BY JURY IN ANY ACTION ARISING OUT OF MATTERS RELATED TO THIS AGREEMENT, WHICH WAIVER IS INFORMED AND VOLUNTARY.
- 11.6. Waiver. The rights of each of the parties under this Agreement are cumulative and may be exercised as often as any party considers appropriate. The rights of each of the parties hereunder shall not be capable of being waived or varied otherwise than by an express waiver or variation in writing. Waiver of any one breach of any provision hereof shall not be deemed to be a waiver of any other breach of the same or any other provision hereof. Failure to exercise or any delay in exercising any of such rights also shall not operate as a waiver or variation of that or any other such right. Defective or partial exercise of any such rights shall not preclude any other or further exercise of that or any their such right. No act or course of conduct or negotiation on the part of any such party shall in any way preclude such party from exercising any such right or constitute a suspension of or any variation of any such right.
- 11.7. Captions. Article captions contained herein are inserted as a matter of convenience and for reference, and in no way define, limit, extend or describe the scope of this Agreement or any provision hereof.
- 11.8. Prevailing Party. If either party hereto brings any action or suit against any other by reason of any breach of any of the covenants, agreements

or provisions of this Agreement, then in such event, the prevailing party, as determined in such action or suit, shall be entitled to have and recover from the other party all costs and expenses of such action or suit, including, without limitation, reasonable attorneys' fees and expenses resulting therefrom, it being understood that the determination of the prevailing party shall be included in the matters which are the subject of such action or suit.

- 11.9. Counterparts. This Agreement may be executed in two or more counterparts, each of which shall be deemed to be an original, but all of which shall constitute one and the same instrument. Upon Purchaser's execution and delivery of this Agreement, this Agreement shall be binding and effective upon, and shall be an irrevocable offer to Seller by, Purchaser until 12:00 noon (New York time) on the next business day following execution and delivery of this Agreement by Purchaser, unless Seller executes and delivers this Agreement prior to such time, in which case this Agreement shall remain binding and effective upon Purchaser in accordance with the terms of this Agreement. This Agreement shall not be binding upon or enforceable against Seller unless and until executed and delivered by Seller and the Escrow Agent has received the Initial Deposit.
- 11.10. Exhibits and Schedules. All Exhibits and Schedules attached hereto are incorporated herein by reference as though fully set forth at length herein.
- 11.11. Notices. (a) All notices, demands, requests, consents and waivers under this Agreement shall be in writing, shall refer to this Agreement and shall be (i) delivered personally, (ii) sent by registered or certified mail, postage prepaid, return receipt requested, (iii) sent by a nationally recognized overnight courier, or (iv) sent by telecopier, with written confirmation of the receipt of such telecopy, addressed as set forth below. If delivered personally, any notice shall be deemed to have been given on the first (1st) business day on or after the date delivered or refused. If mailed, any notice shall be deemed to have been given on the earlier to occur of the first (1st) business day on or after the date of delivery or the third (3rd) business day after such notice has been deposited in the U.S. mail in accordance with this Section 11.11. If sent by overnight courier, any notice shall be deemed to have been given on the first (1st) business day on or after the date following the date such notice was delivered to or picked up by the courier. If sent by telecopier, any notice shall be deemed to have been given (I) on the first (1st) business day on or after the date sent, if confirmation of receipt hereof is given on or before 5:00 p.m. (New York City time), or (II) on the next business day, if confirmation of receipt thereof is given after 5:00 p.m. (New York City time). Copies of all notices shall be given in accordance with the above as follows:

TO SELLER: NBBRE-919 Third Avenue Associates, L.P.

c/o NBB Real Estate Inc.

530 Fifth Avenue

New York, New York 10036 Attention: Toshihiko Asai Telecopier: 813-3281-0215

WITH A COPY TO: Gibson, Dunn & Crutcher, LLP

200 Park Avenue

New York, New York 10166

Attention: Andrew H. Levy, Esq. Telecopier: (212) 351-4035

and

J.P. Morgan Investment Management Inc.

522 Fifth Avenue

New York, New York 10036 Attention: Donald Rederscheid Telecopier: (212) 837-5208

TO PURCHASER: Reckson Operating Partnership, L.P.

c/o Reckson Associates Realty Corp.

10 East 50th Street

New York, New York 10022

Attention: Philip M. Waterman III

Telecopier: (212) 715-6535

WITH A COPY TO: Fried, Frank, Harris, Shriver & Jacobson

One New York Plaza New York, New York 10004

Attention: Joshua Mermelstein, Esq.

Telecopier: (212) 859-4000

- (b) Any counsel designated above or any replacement counsel which may be designated respectively by either Seller or Purchaser or such counsel by written notice to the other party is hereby authorized to give notices hereunder on behalf of its respective client.
- 11.12. Construction. Whenever the singular number is used herein and when required by the context, the same shall include the plural, and the masculine,

feminine and neuter genders shall each include the others, and the word "person" shall include corporation, firm, partnership, joint venture, trust or other entity.

- 11.13. Amendment. This Agreement may be amended or modified only by a written agreement executed by the person against whom enforcement of such amendment or modification is sought.
- 11.14. Entire Agreement. Other than that certain confidentiality letter, dated February 4, 1999, from or on behalf of Purchaser to Seller (the "Confidentiality Agreement"), this Agreement contains the entire agreement of the parties with respect to the subject matter hereof, and no representations, inducements, promises or agreements, oral or otherwise, between the parties not contained herein shall be of any force or effect, all of the same being merged herein.
- 11.15. Governing Law. This Agreement shall be governed by and construed in accordance with the internal laws of the State of New York without regard to principles of conflicts of law.
- 11.16 Monetary Amounts. All monetary amounts referred to herein are in United States Dollars.
- 11.17 Time of the Essence. Time shall be of the essence as to Purchaser's performance of all of its obligations under this Agreement.
- 11.18. Confidentiality. All terms and conditions of the Confidentiality Agreement shall remain in full force and effect according to its terms during the pendency of this Agreement and such terms thereof as are intended to survive acquisition of the Loan by Purchaser shall continue to survive. The provisions of this Section shall survive the Closing or any expiration or termination of this Agreement.
- 11.19. No Recording. Neither Seller nor Purchaser shall record, or arrange to record, this Agreement with any filing office in any jurisdiction.
- 11.20. Review Materials. On or before the Due Diligence Expiration Date, Purchaser and Seller shall jointly deliver to Escrow Agent a box containing all Review Materials as of such date. Escrow Agent shall hold such Review Materials (and any Review Materials thereafter jointly +delivered to Escrow Agent), at Purchaser's and Seller's joint expense, pursuant to the terms of an escrow agreement in form and substance reasonably satisfactory to Purchaser, Seller and Escrow Agent. Seller agrees that, unless requested by Purchaser, Seller shall not deliver (or make available to Purchaser at the room in the Real Property labeled "War Room") any Review Materials not previously delivered or made available to Purchaser during (a) the two (2) business day period prior to the Due Diligence Expiration Date, or (b) the two (2) business day period prior to the Closing Date.
- 11.21. Consent to Jurisdiction. Each of Purchaser and Seller hereby irrevocably and unconditionally (a) submits itself, solely for the purposes of any legal action or proceeding relating to this Agreement or for recognition and enforcement of any judgment in respect thereof, to the non-exclusive jurisdiction of the Supreme Court of the State of New York, the courts of the United States of America for the Southern District of New York, and appellate courts thereof (collectively, the "New York Courts"), (b) consents to the bringing of any such action or proceeding in the New York Courts and waives any objection that it may now or hereafter have to the venue of any such action or proceeding in any such court, including, without limitation any objection that such action or proceeding was brought in an inconvenient court, and agrees not to plead or otherwise assert the same, (c) agrees to service upon it of any and all process in any such action or proceeding at the address and in the manner set forth in Section 11.11 hereof, (d) agrees that nothing herein shall affect the right to effect service of process in any other manner permitted by law, and (e) agrees that a final judgment in any such action or proceeding shall be conclusive and may be enforced in other jurisdictions by suit on the judgment or in any other manner provided by law.
- 11.22. Acquisition of Partnership Interests. Purchaser has requested that Seller consider arranging for the sale of all of the partnership interests in Seller rather than a sale of the Loan. Seller shall consider such request once Purchaser delivers to Seller a written proposal regarding such sale; provided, however, that (a) neither Seller's consideration of such proposal nor anything contained herein shall be deemed a binding agreement to sell partnership interests in Seller, (b) Seller may accept or reject, in its sole and absolute discretion, any proposal by Purchaser, and (c) no person shall be bound to sell any partnership interests unless and until definitive documentation is executed and delivered by all parties thereto.

IN WITNESS WHEREOF, the parties have executed this Agreement as of the date first above written.
SELLER:
NBBRE-919 THIRD AVENUE ASSOCIATES, L.P.
By: NBB Real Estate, Inc., its General Partner
By: Name: Title:
PURCHASER:
RECKSON OPERATING PARTNERSHIP, L.P.
By: Reckson Associates Realty Corp. its General Partner
Ву:
Scott Rechler President
Employer Identification Number:
THE UNDERSIGNED JOINS IN THE EXECUTION HEREOF SOLELY FOR THE PURPOSES OF AGREEING TO ACT AS "ESCROW AGENT" PURSUANT AND SUBJECT TO THE PROVISIONS OF THIS AGREEMENT:

GIBSON, DUNN & CRUTCHER LLP

Andrew H. Levy, Partner

RECKSON OPERATING PARTNERSHIP, L.P. c/o Reckson Associates Realty Corp. 10 East 50th Street New York, New York 10022

As of May 10, 1999

NBBRE-919 Third Avenue Associates, L.P. c/o NBB Real Estate, Inc. 530 Fifth Avenue
New York, New York 10036

Re: 919 Third Avenue (the "Property")

Gentlemen:

Reference is made to that certain Purchase and Sale Agreement (the "Purchase Agreement") by and between Reckson Operating Partnership ("ROP") (ROP, together with its affiliates, collectively, "Reckson") and NBBRE-919 Third Avenue Associates, L.P. (the "Lender") dated as of May 10, 1999, with respect to the sale by Lender to ROP of the mortgage loan as more particularly described in the Purchase Agreement (the "Transaction").

- 1. This letter confirms the agreement of Reckson and the Lender that, as of the Closing Date, with the approval of ROP, all of the leasing commissions listed on Schedule 7 shall have been paid, and at the Closing, Lender shall deliver to ROP a receipt from ESG (as defined in the Purchase Agreement) evidencing such payment.
- 2. This letter confirms the agreement between Reckson and the Lender that, in addition to any restrictions set forth in the Confidentiality Agreement (as hereinafter defined), any future press release and/or public disclosure of information with respect to the Transaction (including, without limitation, any filing with the Securities and Exchange Commission), if any, issued and/or disclosed by Reckson shall be delivered to the Lender, for the Lender's review (and, solely with respect to any aspect of such press release and/or public disclosure of information which is not required to be issued and/or disclosed by law, rule or regulation, approval) at least twenty-four (24) hours before the issuance of any such press release and/or disclosure of such information. To the extent required pursuant to the preceding sentence, the Lender's approval shall not be unreasonably withheld, and shall be deemed granted if (a) not denied in writing received by Reckson within twenty-four (24) hours after the proposed press release and/or public disclosure of information is received by the Lender (time being of the essence), or (b) the proposed press release and/or public disclosure of information is substantially similar to a press release or public disclosure previously approved by Lender; provided, however, that the terms of clause (b) shall not

apply to the initial press release and "8-K" to be filed immediately after the Closing Date. Any such denial shall identity the portions of such press release which are objected to by the Lender.

- 3. This letter confirms the agreement of Reckson and Lender that attached hereto as Exhibit "A" is a true, correct and complete copy of that certain confidentiality letter, dated as of February 9, 1999 (the "Confidentiality Agreement"), executed on behalf of Reckson for the benefit of the Lender, Borrower and ESG.
- 4. This letter confirms the agreement of Reckson and the Lender that the Purchase Agreement is hereby amended as follows:
- (a) Section 11.20 of the Purchase Agreement is hereby amended to delete the first two (2) sentences thereof in their entirety and substitute the following in its place:

"On or before the Due Diligence Expiration Date, the representatives of both Purchaser and Seller shall jointly identify (by initialing a copy of an index of Review Materials) and make two (2) copies of the Review Materials (other than building plans) as existing as of the date two (2) business days prior to the Due Diligence Expiration Date, and shall place a complete set of such Review Materials in two separate boxes, one to be held by Purchaser, and one to be held by Seller (the "Boxes"). With respect to building plans, Purchaser agrees (a) to cause a copy of the building plans as existing as of the Closing Date to remain at the Property, (b) that Seller shall have the right, from time to time during business hours and upon reasonable notice to Purchaser, to inspect the building plans, and (c) that if Seller requests copies of all or any portion of the building plans, Purchaser shall deliver such copies to Seller and the cost of such copies shall be borne equally by Purchaser and Seller (the terms if this sentence shall survive the Closing). Subject to the terms of the next immediately succeeding sentence, after the Due Diligence Expiration Date and prior to the Closing Date, representatives of Seller and Purchaser may, from time to time, jointly identify (by

(b) Section 6.5(e)(ii) of the Purchase Agreement is hereby amended to delete the last two (2) sentences thereof in their entirety and substitute the following in their place:

"With respect to real estate tax refunds or credits with respect to the Real Property relating to the fiscal year in which the Closing occurs or relating to prior fiscal years, (a) to the extent that Seller has authority, Seller hereby authorizes and directs such refunds or credits to be paid to the Cash Collateral Account (it being understood and agreed that Seller makes no representation or warranty regarding its authority to so authorize and direct such payment), and (b) as and when such refunds or credits are paid, Purchaser shall pay to Seller an amount equal to the excess, if any, of any such refunds or credits, over the amount of such refunds or credits paid or required to be paid to tenants (it being understood and agreed that Purchaser shall pay to Seller all refunds or credits required to be paid to tenants which are not actually paid to tenants). The terms of this Section 6.5(e) shall survive the Closing."

5. Except as modified and amended by this letter, all of the terms, covenants and conditions of the Purchase Agreement are hereby ratified and confirmed.

Kindly acknowledge your agreement to the foregoing by executing four (4) copies of this letter where indicated and then return two (2) fully executed copies to the undersigned.

Very truly yours,

RECKSON OPERATING PARTNERSHIP, L.P.

By: Reckson Associates Realty Corp. its General Partner

Bv:

Name: Jason Barnett

Title: EVP and General Counsel

AGREED AND ACCEPTED AS OF THIS 10TH DAY OF MAY, 1999:

NBBRE-919 THIRD AVENUE ASSOCIATES, L.P.

By: NBB Real Estate, Inc.

its General Partner

ву:

Name: Ichiro Tsunada Title: President Reekson Announces Closing on \$277.5 Million Acquisition of First Mortgage Note Secured by 919 Third Avenue in New York City

(MELVILLE, NEW YORK, June 15,1999) - Reekson Associates Realty Corp. (NYSE: RA) has announced that the Company has closed on the acquisition of the first mortgage note secured by 919 Third Avenue, for a purchase price of approximately \$277.5 million. The 919 Third Avenue office tower is a 1.4 million square foot, 42 story, Class A office building located at Third Avenue between 55th and 56th Street in New York City.

Reckson's acquisition of the first mortgage note secured by 919 Third Avenue entities the Company to all of the net cash flow of 919 Third Avenue and to substantial rights with respect to the operation of the property, with the Company anticipating ultimately obtaining title to the property. This transaction is Reckson's first New York City real estate acquisition subsequent to the Company's acquisition of Tower Realty Trust and increases the Company's presence in New York City to approximately 3 million square feet.

Scott Rechler, Reckson's Co-Chief Executive Officer and President commented "This transaction continues our value creation strategy into the Manhattan marketplace and demonstrates our ability to source opportunities, create innovative structures and complete transactions in the New York City market."

The property is expected to initially generate an NOI yield of 9.5% until approximately 705,000 square feet is vacated in April of 2000. This space has been re-leased to three high quality tenants. Additionally, Reckson will complete an \$90 million repositioning program, including tenant build-out, that is already underway. Once these new tenants take occupancy in the first quarter of 2002, the building will begin to re-stabilize at an approximate 9.5% NOI yield. The yield is expected to increase to 11.5% as below market leases are replaced through 2005.

Reckson is purchasing the first mortgage note for approximately \$200 per square foot. After the completion of the repositioning and re-leasing of the vacated space the Company anticipates a total investment cost of \$275 per square foot, equivalent to an approximate 35% discount to replacement cost.

Tod Waterman, Managing Director of Reckson's New York City Division, said, "This transaction, involving one of the premier Class A office towers in Manhattan, marks Reckson's strategic entry into the Third Avenue corridor. We now have assets in four key Manhattan submarkets which will allow the Company to participate in the continued strengthening of the New York City market,"

Reekson Associates Realty Corp. is a self-administered and self-managed real estate investment thist (REIT) specializing in the acquisition, leasing, financing, management and development of office and industrial properties.

Reckson's core growth strategy is focused on the markets surrounding and including New York City. Since the completion of its initial public offering in May 1995, Reckson has acquired, contracted to acquire or developed approximately \$2 billion of real estate assets comprising approximately 20.5 million square feet of space.

Reckson is one of the largest publicly traded on-,ners and managers of Class A office and industrial properties in the New York City "Tri-State" area, with 210 properties comprised of approximately 25 million square feet either owned and controlled directly or indirectly, or under contract.

This Information contains forward-looking information that is subject to certain risks, trends and uncertainties that could cause actual results to differ materially from those projected. Among those risks, trends and uncertainties are the general economic climate; the supply of and demand for office and industrial properties in the New York metropolitan Tri-State area: interest rate levels; the availability of financing, and other risks associated with the development and acquisition of properries, including risks that development may not be completed on schedule, that the tenants will not take occupancy or pay rent, or that development or operating costs maybe greater than anticipated For further information in factors that could impact Reckson, reference is made to Reckson's filings with the Securities and Exchange Commission.